

Temi di discussione

(Working Papers)

Firms' innovation and university cooperation. New evidence from a survey of Italian firms

by Daniela Bragoli, Flavia Cortelezzi and Massimiliano Rigon





Temi di discussione

(Working Papers)

Firms' innovation and university cooperation. New evidence from a survey of Italian firms

by Daniela Bragoli, Flavia Cortelezzi and Massimiliano Rigon

Number 1400 - February 2023

The papers published in the Temi di discussione series describe preliminary results and are made available to the public to encourage discussion and elicit comments.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

Editorial Board: Antonio Di Cesare, Raffaela Giordano, Monica Andini, Marco Bottone, Lorenzo Braccini, Luca Citino, Valerio Della Corte, Lucia Esposito, Danilo Liberati, Salvatore Lo Bello, Alessandro Moro, Tommaso Orlando, Claudia Pacella, Fabio Piersanti, Dario Ruzzi, Marco Savegnago, Stefania Villa. *Editorial Assistants:* Alessandra Giammarco, Roberto Marano.

ISSN 2281-3950 (online)

Designed by the Printing and Publishing Division of the Bank of Italy

FIRMS' INNOVATION AND UNIVERSITY COOPERATION. NEW EVIDENCE FROM A SURVEY OF ITALIAN FIRMS

by Daniela Bragoli*, Flavia Cortelezzi** and Massimiliano Rigon***

Abstract

In this paper, we investigate whether cooperation with universities stimulates the innovative performance of Italian firms. We use a dataset merging information from two different surveys carried out by the Bank of Italy between 2007 and 2010. We derive our findings from a two-stage procedure in order to rule out any spurious correlations resulting from omitted variables. The results show that cooperation with universities does not affect the likelihood of firms introducing technological innovations. However, when we distinguish between pure technological innovation outcomes (only new products and/or productive processes) and joint innovation outcomes, which involve both organizational and technological changes, we find that only the latter are positively stimulated by cooperation with universities. These findings are promising since, according to the innovation management literature, joint innovation activities are more successful in translating new ideas and new business opportunities into market success.

JEL Classification: C35, C36, O30.

Keywords: university cooperation, technological innovation, organizational innovation, control function.

DOI: 10.32057/0.TD.2022.1400

Contents

1.	Introduction	5
2.	Literature review	9
	2.1 Technological and organizational innovation: the importance of combining them	12
3.	Data	. 14
	3.1 Data sources	. 14
	3.2 Organizational, technological, joint innovation	. 15
	3.3 Cooperation with universities	. 16
	3.4 Control variables	. 16
	3.5 Instruments	. 17
	3.6 Descriptive statistics	20
4.	Empirical strategy	22
5.	Results on the link between cooperation and innovation	24
	5.1 The binary model	25
	5.2 The multivariate model	. 30
6.	Concluding remarks	. 33
	pendix	
Re	ferences	. 37

^{*} Department of Mathematics for Economic, Financial and Actuarial Sciences, Università Cattolica del Sacro Cuore, Milano, Italy.

^{**} Department of Economics and Law, Insubria University (Como), Italy.

^{***} Bank of Italy, Milano Branch, Economic Research Unit.

1. Introduction¹

Innovation activities are becoming more and more crucial for the economic survival of firms. Indeed, rapid technological changes, shorter product life cycles and globalization processes, have deeply transformed the competitive environment, making it necessary for firms to develop new products, to change productive processes or to implement new technologies (OECD, 2018). According to the literature on the knowledge-creating view of the firms, one possible channel to enhance innovation is increasing firms' access to external knowledge, to integrate and complement internal ideas and technologies with outside information (Cohen and Levinthal, 1990; Amesse and Cohendet, 2001; Chesbrough, 2003; Caloghirou et al., 2004). Within today's knowledge-based societies, universities play a crucial role since they represent an invaluable source of qualified knowledge, creating and disseminating new expertise and inventions, through basic and applied research (Jaffe, 1989; Mansfield, 1991; Mansfield and Lee, 1996; Etzkowitz and Leydesdorff, 1997; Mowery and Sampat, 2005).

In the mid-nineties the European Commission labelled as the "European Paradox" the perceived failure of European countries to translate scientific advances into marketable innovations (European Commission, 1995). The mismatch between the high level of scientific results achieved in research centers or universities and the low degree of competitiveness of firms in Europe, suggested that the interchange between firms and universities was superficial and should have been incentivized by policymakers. Since then, national governments and regional authorities tried to encourage technology transfer activities from universities to firms (see Kochenkova et al. 2016 for a survey on the role of governments in facilitating the commercialization of academic research and university–firm collaborations).

Economic theory recognizes a mutual benefit of the university-industry cooperation, but also tensions that may arise between the two institutions. Universities might be interested in the cooperation with firms to obtain funds for research assistance, lab equipment and their own research agenda and to obtain insights into their own research by being able to integrate theory with empirics (Hall, 2004). Similarly, firms might value their relationship with universities because there is some desire of risk sharing and because it gives them an early look-in at discoveries, they may not be able to anticipate (Lee, 2000). However, the specific characteristics of scientific knowledge, which involve technological innovations, make the

¹ We are grateful to Luca Citino, Luigi Infante, Giovanni Marseguerra, Sara Pinoli, Paola Rossi and seminar participants at the R&D Management Conference at the University of Cambridge (UK) for extremely helpful and insightful comments. Daniela Bragoli and Flavia Cortelezzi have collaborated in the development of the economic framework. Statistical treatment of the data was carried out by Massimiliano Rigon. The opinions expressed are those of the authors and do not reflect the view of the institutions to which they are affiliated. All remaining errors are our own.

cooperation between universities and firms for R&D projects difficult. This type of cooperation is in fact characterized by high uncertainty, high information asymmetries between partners, high transaction costs for knowledge exchange, requiring the presence of absorptive capacity at each side of the market transfer and high spillovers to other market actors (Hall, 2004; Hall et al. 2001, 2003; Cassiman and Veugelers, 2005). It follows that there is not a full agreement on the effects of university-industry cooperation; whether more innovative firms are more prone to cooperate with external partners and whether this cooperation is beneficial for technological innovation is not well defined even in the empirical literature.²

Our work is related to this strand of empirical papers with some novel aspects. First, we analyse the behaviour of Italian firms. The case of Italy is interesting since, as discussed in Bugamelli et al. (2012), starting from the '90 the productivity of Italian firms has slowed down compared to other advanced economies and the technological gap has widened given the low propensity of Italian firms to innovate. The literature on the topic has identified three main drivers to the gap: the specialization in traditional sectors, the firms' governance, often based on family control, and firms' size.³ The latter, which is strongly linked to the others, makes it very hard for Italian firms to specialize in high and intensive knowledge sectors, which are characterized by relevant economies of scale, oligopolistic competition and by amounts of investments which are larger than those typical of Italian firms.⁴ Small and medium Italian firms usually do not have internal research labs and finance their R&D activities through internal funds (Bragoli et al. 2016; Bragoli et al. 2020; Barbieri et al. 2020). It follows that the cooperation with other firms and with universities or research centers could become crucial in order to bridge – at least partially – the technological gap of Italian firms with foreign competitors.⁵ The contributions on university-industry cooperation are summarized for the Italian case in Grimaldi et al. (2021) for what concerns regional specialization, whereas Bellandi et al. (2021) put forward an interpretative framework that links central entrepreneurial or engaged strategies with the way academics organize their third mission activities. Other studies, concentrating on the specificities of the Italian industrial

² See for example literature on complementarity and substitution between internal and external sourcing (Cassiman and Veugelers, 2002; Cassiman and Veugelers, 2005).

³ In Italy, the fragmentation of the production system is greater than in all the other major advanced economies. In terms of added value and employment, companies with less than 250 employees represent, respectively, about 70% and 80% of the total, more than 10 percentage points higher than the average of the European Union countries (European Commission, 2017).

⁴ A different stream of the literature has emphasized some positive aspects of the Italian production system, showing that Italian small and medium-sized enterprises, especially manufacturing companies that have been able to exploit agglomeration economies in the bright activity of industrial districts, have themselves been the actors of a success story that has combined growth and innovation for a long time. In spite of the strong competition of less developed countries, Industrial Districts firms have been able to strongly differentiate and increase the added value of their products (Sforzi, 2009 and Marseguerra et al. 2016).

⁵ Using data on 1,500 regions across 78 countries from 1950 and 2010, Valero and Van Reneen (2019) find that increases in the number of universities are positively associated with future growth of GDP per capita. Part of the effect of universities on growth is mediated through an increased supply of human capital and greater innovation.

sector, focus on sectoral effects on the patterns of cooperation, especially in low-tech regions (Parmentola et al., 2021; Maietta, 2015; Giuliani and Arza, 2009).⁶

Secondly, we distinguish between different types of innovation outputs since the OECD and Eurostat have found the need of diversifying between technological and other types of innovation, introducing also organizational and marketing innovations into the guidelines for collecting and interpreting innovation data (Oslo-Manual, OECD and Eurostat 2005). The rationale of this distinction stems from the fact that technological innovation has been criticized for not fully capturing innovation in services and for ignoring important elements of innovative activities of firms, such as the adoption of new practices and the re-organization of existing business routines, external relations, and marketing (Hipp and Grupp 2005; Baranano, 2003; Boer and During, 2001).

In particular, the aim of this paper is not only to provide evidence on whether cooperation with universities encourages innovation, but also to investigate what type of innovation output cooperation with universities targets. We distinguish between three different types of innovation: pure organizational, pure technological (new products or productive processes) and joint innovation which occurs when firms introduce both organizational and technological changes. The conceptual framework which motivates our analysis is linked to the idea of entrepreneurial university (see Cerver Romero et al., 2021 for a recent review), according to which the role of universities has evolved considerably over time from an entity producing 'knowledge for its own sake' to being a source of knowledge that is requisite for economic growth. Within this framework the role of universities goes beyond generating technology transfer, rather the mandate of the university in the entrepreneurial society is to contribute and provide leadership for creating entrepreneurial thinking, actions, institutions and what Audretsch et al. (2006) refer to as entrepreneurship capital. Universities encourage not only technological transfer but also knowledge transfer, making the innovation process more pervasive in nature involving different forms of activities and not only the more traditional purely technological type.

Empirically most papers on the firm-university cooperation focus on the effects on technological innovation, however there is a vast literature - sometimes labelled as "innovation management literature" - that points out different reasons why joint innovation is relevant (see Damanpour, 2020). The first relates to the fact that investing in both types of innovation activities is very common, since some kinds of innovations

⁶ Other studies on the Italian case include Medda et al. (2006) focus on the impact research cooperation on TFP growth. Hall et al. (2013) use different waves of the "Survey on manufacturing firms" conducted by the Italian commercial bank Unicredit to investigate the impact on R&D and ICT investment on innovation and productivity of Italian firms. Differently from our paper, they focus on internal resources of firms and do not analyse the interaction with universities.

do not have exclusively a technological or organizational content. Innovations influence each other and hence, they ought to be implemented in conjunction (Hervas-Olivier et al. 2012; Johannessen et al., 2001; Drejer, 2002; Garcia and Calantone, 2002). Secondly, the presence of a joint strategy, which goes beyond the introduction of new products and services but also pervades the whole economic system, involving also new business and management models, might be related to a type of innovation that requires a more intense change, for which the complementarities between university and firms could be stronger (Ringberg et al., 2019). As a third aspect, the innovation management literature stresses the importance of integrating both types of innovative activities for successfully transferring new ideas and new business opportunities into market success (Tidd et al., 2001; Cozzarin and Percival, 2006 and 2008; Evangelista and Vezzani, 2010).

We depart from most of other papers on the topic, which rely on multivariate binary models and simultaneous equations to account for reverse causality. In contrast, we use a two-stage econometric strategy. More precisely, since the model is nonlinear, we derive our results using a 'two-stage residual inclusion' procedure as in Terza et al. (2008) and Wooldridge (2014 and 2015).⁷ To rule out spurious correlations, due to the possible existence of omitted variables, we use as exclusion restriction in the first stage a variable which combines the distance between firms and universities with a measure of universities' openness to external cooperation.

We use a unique Italian firm level database merging two different datasets, which contain data on Italian firms with more than 19 employees collected by the Bank of Italy through two different surveys: the Survey of industrial and service firms and the Bank of Italy's Business Outlook Survey of industrial and service firms. Results show that the cooperation with universities increases the probability of a joint innovation activity, while we do not find any significant effect when we consider a broader definition of technological innovation which encompasses both pure technological and joint innovation. This finding implies that universities become crucial when the innovation is not only the creation of a new product or productive process, but when it also introduces changes that pertain to different spheres of the firms' activities bringing it closer, to some extent, to the Schumpeterian concept of disruptive innovation. This is good news since, according to the innovation management literature, these innovations can be more easily translated into market success. It worth noticing that the last available dataset provided by the Innovation Community Survey (CIS) shows that the share of innovative firms collaborating with universities or public

⁷ When we consider the binary outcome (according to the introduction or not of a technological innovation) we also estimate a linear probability model to compare the results with those obtained with the residual inclusion technique, as described in the section 4.

research centers is still low⁸. Therefore, despite of the fact that our results refers to the past decade, they still contain valid policy implications: the firm-university cooperation should be encouraged to stimulate successful innovation and increase the productivity of Italian firms, thus reducing the technological gap with their competitors.

The rest of the paper is organized as follows. Section 2 outlines the literature review on the topic; Section 3 provides an overview of the data and delineates the variable selection process; Section 4 describes the econometric strategy; Section 5 contains the main results; finally, Section 6 concludes.

2. Literature Review

The literature that we review in this Section can be divided in two main strands. One is mainly interested in studying the complementarity/substitution between internal and external sources of knowledge, the other is instead mainly focused in analysing the effect of external cooperation on innovation.

According to the first stream, whether outside sources of knowledge are important for the firm to generate innovation depends on how the external knowledge interacts with its internal capabilities, i.e., with the accumulated knowledge and skills that enable the firm to coordinate its activities and deploy resources advantageously.

An extensive theoretical literature in Industrial Organization has developed around the effects of external spillovers on own R&D (see De Bondt, 1996 for a review). According to some contributions, which focus on R&D spillovers among firms, not only external R&D will typically substitute for own R&D in the receiving firm, but spillovers will also reduce own R&D by the sending firm, since the latter cannot fully internalize all benefits from its investment (Spence, 1984). To the contrary, according to other studies, a firm's ability to identify, assimilate and exploit existing external technologies can be enhanced by own R&D (e.g., Harabi, 1995). The notion of absorptive capacity introduced by Cohen and Levinthal (1990) stresses the importance of a stock of prior knowledge to effectively absorb spillovers. In this last setting, the desire to assimilate external know-how creates a positive incentive to invest in R&D. Hence spillovers, rather than diminish own R&D, may encourage industry R&D investments.

The empirical literature that follows this first stream has taken different directions. Some studies address the relation between internal and external R&D investigating the role of R&D cooperation with external partners (Veugelers, 1997; Cassiman and Veugelers, 2002; Cassiman and Veugelers, 2005), such as

⁸ According to the CIS 2018 (the last publicly available) the share of Italian innovative firms (as defined in the Oslo manual 2018, whose definition of innovation is broader than the one used in our paper) increased over the 50% mark; however only about 11% of these innovative firms collaborated with universities or research centers

other firms (Arvanitis et al., 2020) or research institutes/universities (Becker, 2003, Becker and Dietz, 2004) and look at the role of complementarity and substitution between internal and external sources of knowledge.

Focusing on the firm-university cooperation, some studies find a complementarity between internal and external sources of knowledge. Becker (2003) shows that, through cooperation, firms improve their technological know-how and internal skills, which increase firms' internal capabilities for developing new products and technologies.

Becker and Dietz (2004) investigate the complementarities between internal and external R&D, studying the effect of R&D cooperation on firms' innovation. They find that R&D cooperation is used to complement internal resources in the innovation process, enhancing the innovation input and output measured by the intensity of in-house R&D and by the realization of product innovations. Along the same line, Gómez et al. (2020) show that extramural investments in R&D play a key role in assuring that firms increase their proportion of high novelty products, whereas they are less relevant when dealing with low novelty products.

However, not all studies find a positive relationship between R&D cooperation and firms' innovation performance. These findings are motivated by a substitutive relationship between specific innovation activities or knowledge sources, which may be explained in the context of path dependencies, switching costs and/or diseconomies of scope (Hagedoorn and Wang, 2012). Some scholars provide empirical evidence on offsetting (or non-synergistic effects) of internal and external knowledge sources on firm innovation (Berchicci, 2013; Laursen and Salter, 2006; Vega-Jurado et al. 2009). More recently, Caloghirou et al. (2021) show an inverted U-shaped relationship between industry-university cooperation and product innovation and show that knowledge stocks (internal capabilities) moderate the relationship between industry-university collaboration and product innovation, suggesting that firms with low levels of knowledge stocks (age, export and employee's educational level) benefit more in terms of innovation from the development of knowledge flows with universities.

The second stream of the literature analyses the relationship between firm-university cooperation and innovation, disregarding the interaction between internal and external sources of knowledge. Indeed, these studies are mainly interested in measuring the additional effects of firm-university cooperation on firms' innovative inputs, outputs or more generally on firm performance, controlling for firms' internal capabilities.

Regarding the effects of university cooperation on firms' innovation output, the first papers that address the issue empirically are those of Jaffe (1989) and Mansfield (1995), both related to the US. The latter shows that in a random sample of 76 large American firms belonging to seven manufacturing industries,

about 10% of the product and process innovations could not have been developed without recent academic research. As in Mansfield (1995), most of the papers identify as innovation the introduction of a new product or a new productive process. Empirical evidence has been provided for firms in different countries and results are mixed. Some papers find positive effects on firms' innovative activities, including R&D intensity (Scandura, 2016), the propensity to register new patents as well as the introduction and sales of new products (Arvanitis et al., 2008; Fritsch and Franke, 2004; Aschhoff and Schmidt, 2008). For others, the results are less clear-cut. Monjon and Waelbroeck (2003) find that cooperation with foreign universities increases the probability of radical innovation. Rouvinen (2002), focusing on Finnish firms, shows that cooperation with universities has a positive impact on product innovation, but no impact on process innovation. Robin and Schubert (2013) find similar results for France and Germany. Lööf and Broström (2008), using Community Innovation Survey (CIS) data on Swedish firms, show that university collaboration positively influences product innovation sales as well as the propensity to apply for patent for manufacturing firms with 100 or more employees, whereas they find no effect for firms in the service sector.

Analysing the impact of university cooperation on the economic performance of firms (i.e. labor productivity, total factor productivity), Medda et al. (2006) do not find any significant effect of collaborative research undertaken by Italian manufacturing firms and universities on total factor productivity growth, whereas other studies find a positive effect on several measures of labor productivity, sales productivity or sales growth (see e.g. Belderbos et al., 2004; Branstetter and Ogura, 2005).

More recently the literature has focused on the role of entrepreneurial university (see Cerver Romero et al., 2021 for a recent review). According to Audretsch (2014) the role of universities has evolved considerably over time from an entity producing 'knowledge for its own sake' to being a source of knowledge that is requisite for economic growth. The emergence of the entrepreneurial university gave universities a dual mandate, i.e. to produce new knowledge but also to facilitate the transfer of technology and knowledge spillovers (Acs et al., 2009; Audretsch, 2014; Link and Sarala, 2019). Universities, through third mission activities, act as agents for stimulating economic growth and development (Rippa and Secundo, 2019) creating new learning environments and representing a context for knowledge spillover creation. Secundo et al. (2021) investigate the role of knowledge spillover entrepreneurship analysing the process of intentional and unintentional knowledge flows among universities, industries, and institutions. Aksoy et al. (2022) base their analysis on the type of knowledge that may influence the success rate of knowledge transfer. In particular, the diversity of the knowledge base and cognitive proximity between partners are found to be important for this purpose.

This recent literature in studying the relation between university and firms has moved away from the notion of 'technology transfer' towards the idea of 'knowledge transfer' which, in turn, is being challenged by the concept of the more free-flowing multidimensional 'knowledge exchange' between the three sectors of the 'triple helix', comprising universities, business and the government (Trencher et al., 2014).

Within this context the role of universities goes beyond generating technology transfer, rather the mandate of the university in the entrepreneurial society is to contribute and provide leadership for creating entrepreneurial thinking, actions, institutions and what Audretsch et al. (2006) refer to as entrepreneurship capital.

Building around the conceptual framework of knowledge transfer in this paper we study the role of university firm cooperation distinguishing between pure technological innovation (new products and/or new productive process), organizational innovation and the joint innovation activity. From our analysis we expect that, since universities encourage knowledge transfer together with technological transfer, the most probable form of innovative activity promoted by cooperation should thus be the presence of both technological and organizational innovation.

In the following subsection, we present the literature, which focuses on the distinction between technological, organizational, and joint innovation.

2.1 Technological and Organizational innovation: the importance of combining them.

Many researchers have used the term innovation to identify technology-based product and process innovations (see Crossan and Apaydin, 2010 for a review). However, while the importance of technology and technological innovation for firms is undeniable, the concept of innovation is broader, and technological innovation that firms generate or adopt (Damanpour, 2020).

The distinction between the different types of innovations dates to Schumpeter, which identifies five different types: new products, new methods of production, new markets, new sources of supply, and new ways to organize businesses. More recently Meeus and Edquist (2006) distinguish between two types of product innovations (innovation in goods and services) and two types of process innovations (technological process and organizational). Tether and Tajar (2008) base their analysis on three dimensions of change: 1) changes to what the firm produces versus changes to how the firm operates, 2) changes to physical technologies versus changes to social technologies, 3) the locus of change, intrafirm versus interfirm. Damanpour (2020) follows the same classification as in Tether and Tajar (2008) but adds a new dimension distinguishing between radical versus incremental innovation.

It is useful to distinguish between different types of innovation since, although each innovation has different characteristics, all of them contribute to firms' performance. Technological innovations are directly related to the main activities of the firm, whereas organizational affect mainly its management systems (Daft, 1978; Damanpour & Evan, 1984). They are processes and practices that modify the way management is conducted and constitute the rules and routines through which tasks are organized within the firm (Birkinshaw et al., 2008; Hamel, 2006). The concept of organizational innovation is complex and difficult to measure. In a literature review, Černe et al. (2016) identify several non-technological innovations including organizational innovation.

In general, while organizational (non-technological process) innovations are recognized as relevant (Arrow, 1962; Sanidas, 2005), and their introduction is considered necessary to renew organizational strategy, structure, and systems (Nickell et al., 2001; Birkinshaw et al., 2008; Volberda et al., 2013), research on this type of innovation lags behind technological innovation, and its effect on performance is considered to be less evident.⁹ Theoretical arguments, however, claim that the combination of the two types of innovation is more effective in terms of performance outcome rather than the single innovative activities.

According to some authors (Damanpour et al., 2009; Roberts and Amit, 2003), innovative performance calls for the introduction of different types of innovations to help adapt organizations to external and internal changes. Innovation types are related to one another, the introduction of one type could favor the introduction of another, and an understanding of the contributions of each type requires an understanding of its relations with the other types (Damanpour, 2020).

Georgantzas and Shapiro (1993) define 'synchronous innovation' the adoption of congruous technological and organizational innovations. They examine the influence of synchronous innovation on performance and find that the independent effect of each innovation type on performance is very small without synchronous innovation. Roberts and Amit (2003), analysing the effect of three types of innovations (product, process, and distribution) on performance in retail banking organizations, find that long-term performance depends on the history of firms innovation activity rather than on the occasional positive result of single innovations (Roberts and Amit, 2003).

Empirical evidence from these studies challenges the notion that firm performance is promoted by the success of a specific type of innovation, whether product, service, process, technical, or managerial.

⁹ Walker et al 2015 integrated the empirical findings from 44 articles published in peer-reviewed journals and found that the adoption of organizational innovations positively affects organizational performance. A comparison of a matched sample of associations of technological and managerial innovations with organizational performance showed that the two types of innovations affect performance similarly.

Instead, they suggest that performance requires harmonious adjustments of various organizational subsystems through the introduction of complementary innovation types (Ballot et al. 2015; Battisti and Iona, 2009; Hervas-Oliver and Sempere-Ripoll, 2014; Naranjo-Gil, 2009). Ringberg et al. (2019) find that joint technological and organizational innovation might involve a more intense change. Some papers have pointed out that the effects of a joint innovation are usually more successful in transferring new ideas and new business opportunities into market success (Tidd et al., 2001; Cozzarin, 2017; Cozzarin and Perzival, 2006 and 2008; Evangelista and Vezzani, 2010).

A possible reason for this is related to the fact that the two innovative strategies (technological and organizational) might be complementary and thus increase the complexity within the firm that, together with internal routines, creates a competitive advantage because the use of a complex strategy may act as a barrier to potential imitators (Rivkin, 2000).

It follows that organizational innovation, by affecting technological innovation positively (Schmidt and Rammer, 2007; Bartoloni and Baussola, 2018), may have an impact on firms' productivity and profitability.

3. Data

3.1 Data Sources

The dataset used in the analysis is the joint result of two different surveys carried out by the Bank of Italy: the Bank of Italy's Survey of industrial and service firms (Invind, hereafter) and the Bank of Italy's Business Outlook Survey of industrial and service firms.¹⁰

Invind represents one of the richest sources of information at the firm level for Italy. It is carried out by the Bank of Italy once a year between March and April since 1984. It collects both quantitative and qualitative information about industrial and service firms. Similarly, the Business Outlook is carried out once a year, between September and October since 1993, on the same sample of firms interviewed with Invind. Information is collected through telephone interviews. In both cases interviewers are officials of the Bank

¹⁰ The database Invind goes back to 1984. The questionnaire is sent to each enterprise at the beginning of each year and the questions refer to the last two years (this allows data consistency to be checked over time). The sample is stratified according to three criteria: sector of economic activity, size and geographical location. The presence of outliers and missing data within the sample is dealt with by means of appropriate statistical techniques. The Business Outlook is similar to Invind but it contains less questions and is more focused on the business cycle.

of Italy, who tend to establish long-run relationships with firms' managers and are also responsible for the accuracy of the collected information.

Both surveys contain questions that are fixed over time, mainly concerning sales, employment and investments, and a series of questions that vary each year according to the specific issues investigated. For this reason, we focus on the 2007 release of the Business Outlook and on the 2010 release of Invind.¹¹ The former contains a section focusing on the cooperation between firms and universities, while in Invind there are some questions on the innovative attitude of firms. In 2007, the Business Outlook collected answers of 4,196 firms, while Invind with the 2010 release interviews 3,937 firms. Unfortunately, in the 2010 release, questions on innovative attitudes were asked to a random subsample of half of the firms. For this reason, the number of available information drop to 1,964. Due to the presence of missing observations, when we merge the two datasets, we end up with a sample of 1,125 firms, which is the dataset used in our empirical investigation. The Bank of Italy surveys have two main advantages. First, they contain high quality data, mainly due to the rigorous procedure followed in the collection. Second, the surveys mainly focus on medium and large firms, which are the most innovative.¹²

3.2 Organizational, Technological, Joint Innovation

In this paper, the firm's innovative strategy is measured using Invind 2010 question, which asks firms whether, between 2008 and 2010, they were successfully involved in one or more of the following activities: product innovations, innovation of productive processes, managerial or organizational innovations. Firms' tasks that involved only managerial or organizational changes are defined as organizational, firms' activities that involved only product and/or process and both product/process are defined as technological, finally firms' activities which involve both technological and organizational innovation are defined as joint. Thus, we use this information to define the dependent variables to investigate the role of university cooperation. Therefore, this paper is mainly focused on technological innovations, distinguishing between those that have been introduced together with organizational changes from those that have not required changes in the

¹¹ A more recent version of the dataset was not possible to consider since only those releases contain the information needed to conduct the analysis on the relation between university-firm cooperation and innovation. Although in the last ten years the relationship between universities and firms may have evolved, we feel that the dataset can still be effectively used to capture the causal relationship between the cooperation with universities and the innovation strategies of firms. The issue we address has in fact a long-run (structural) nature.

¹² In this paper we do not consider young innovative companies and start-ups. These firms are usually very small and risky. They are focused on the development and the implementation of new technologies, and they deserve and *ad hoc* analysis (see, e.g., Pellegrino et al., 2012).

organization of the firms. We use three different definitions of dependent variable. Two are dummy variables. The first takes value equal to 1 when the firm has introduced a technological innovation (either pure or joint) and zero otherwise; alternatively, we define the dependent variable distinguishing the firms that adopted a joint innovation strategy from the others. Finally, we construct a factor variable, which can take four different values (0=no innovation, 1=pure organizational, 2=pure technological, 3=joint), with the aim of studying whether the firm university cooperation targets a specific type of innovation activity.

3.3 Cooperation with Universities

The main regressor captures the collaboration between the firm and academic institutions. The variable is based on two questions contained in the 2007 release of the Business Outlook. Firms are asked whether they collaborate with one or more universities and in case of positive answer whether the collaboration agreement involves 1) the financing of R&D projects or cooperation in a research project; 2) the purchase of consulting services; 3) a university students' internship in the firm.

We use the answers to these questions to build a binary variable (*university*), which takes the value equal to 1, when the cooperation between firms and universities involves the purchase of consulting services or the financing of R&D projects. When firms and universities do not cooperate or when the collaboration implies only student internships in the firms, then the variable university takes value equal to 0. We make this choice because we expect that the innovation activity is the outcome of a close interaction between firms and universities; in contrast, 'soft' types of cooperation, such as students' internships, most likely will not provide useful contribution to firms' innovation.

3.4 Control variables

We introduce several controls to account for different firms' characteristics and to reduce the omitted variable bias. Table A1 in the appendix shows the complete list of the variables used in the analysis. For the sake of clarity, we group these variables in three vectors. Vector B includes balance sheet variables, namely sales and investments. These variables are computed as the average value of the three years preceding the collaboration period investigated in the survey and they enter the model in natural logarithms. The vector F includes other information related to firms: size (*size*) measured by the number of employees (thousands of units); age (*age*), which considers the number of years since the firm was founded; *group* is a factor variable, which takes values 0 if the firm does not belong to a group, value 1 if the firm is part of an Italian group and value 2 if the group has a foreign ownership; *export* is a categorical variable, which classifies firms in four

different groups: non-exporting firms and firms whose share of sales due to exportations is less than 1/3, between 1/3 and 2/3 and more than 2/3.¹³ The variable *bluecollar* is given by the average share of blue-collar workers over total employees; finally the *dummy district* identifies whether the firm is located in one of the industrial districts defined by the Italian Institute of Statistics.¹⁴

The vector I includes two dummies, which should capture the firm's internal capabilities. The variable *research center* takes value equal to 1 if the firm has a research center (either in Italy or abroad) and zero otherwise, while the dummy R c D distinguishes firms that invest in internal R&D from the others. As shown in Table A2, all these variables are highly correlated with the cooperation dummy confirming the results in the literature, according to which firms are more likely to access to external source of knowledge when they are more able to internalize their positive spillovers.

Finally, we include *sectoral* and *geographical* dummies to consider fixed effects related both to the sector of economic activity of the firm and to its geographical (regional) location.

3.5 Instruments

It is worth noticing that, since collaboration is observed in the period 2005-07, before the innovation is realized (2008-2010), we are sure that our estimates do not suffer from the reverse causality problem. In addition, we must consider that usually there is a systematic delay between the beginning of the collaboration and the observation of the results. Therefore, combining two different releases of Invind and Business Outlook, we can track how collaboration between 2005 and 2007 influences the achievement of innovation in the following three-year period.

However, there is still a possible bias due to omitted variables. The expected sign of the bias is not clear. On the one hand, the estimates could be upward biased since most of the reasons that induce firms to collaborate with universities may be also relevant to explain their innovation activity. On the other hand, more innovative firms may not want to cooperate with universities, since the latter may involve transaction costs and appropriability issues.

To consider possible confounding effects, we have introduced several controls. Some controls are suitable to capture the internal capabilities of the firms i.e., the presence of a research center within the firm

¹³ We have also considered the case in which firms are distinguished in only two groups, exporting and non-exporting firms, respectively. Results, not shown in the paper, are substantially unchanged.

¹⁴ A more detailed description of the Italian Industrial districts is provided in the Istat web page: https://www.istat.it/en/archivio/150367

or the group, and the firm's spending in R&D. Still, we cannot be sure that we have eliminated all the possible endogeneity in the model. Our strategy to tackle the endogeneity issue is to identify an exogenous variable which directly influences firms' cooperation with universities, but, at the same time, it does not directly affect the innovation strategy.

To our knowledge, in the economic literature, suggestions about a suitable instrument for the firmuniversity cooperation seem to be missing. We think that the geographical distance between the firm and the nearest university could be a good instrument for our purpose. Geographical proximity plays a fundamental role as a determinant of the firm-university collaboration, since firms that are located near universities may frequently collaborate with them and benefit from knowledge spillovers (D'Este et al., 2013). Geographical proximity enables the transmission of tacit knowledge, which is personal and context-dependent; it cannot be easily bought via the market, and it is difficult to communicate other than through personal interaction in the context of shared experiences (Morgan, 2004). Geographical proximity matters when knowledge spillovers are informal and in the event of information asymmetry between researchers and research users, which arises when users cannot precisely evaluate the applicability of the transferred research until they attempt to translate it into new or improved products or processes (Audretsch and Feldman, 1996; Landry et al., 2007). In the context of asymmetry, the transfer of knowledge is unlikely if researchers and research users do not have frequent interactions. The number of universities within the region in which a firm is located also affects the probability of interacting with a nearby university because it increases the range of options that are available to a firm (D'Este and Iammarino, 2010).

Fantino et al. (2015) show that in Italy the distance affects the probability of collaboration between firms and universities. Their results are in line with other studies on the role of firm-university geographical proximity (see, for example, Laursen et al., 2008 and Hewitt-Dundas, 2013). Maietta (2015), focusing on the Italian food and drink industry, finds that the probability of cooperation between firms and universities reduces as their distance increases.

In our paper, we compute the geodesic distance between the Italian universities and all firms in the Invind archive. We consider the firm's address and in the case of universities with two or more campuses we consider the distance between each firm and all the university's campuses. However, we are aware that it is possible that the distance itself can be considered endogenous, depending on the firm's location choice: firms can decide to locate close to a university if they are likely to cooperate with that university. Indeed, most of Italian universities are older than firms in our sample. Although this endogeneity problem might still be present, we think that it is more relevant for start-ups, which may be universities spin-offs and are generally founded by researchers or in collaboration with researchers of the same university. Our sample focuses on firms with at least 20 employees and start-ups are not included in the analysis. Nevertheless, to reduce the likelihood that the restriction used in the first stage might be loosely endogenous, we combine the geographical distance with a measure of university openness to external collaboration. For this purpose, we use the results of the first Anvur¹⁵ report on the so-called 'third mission', which contains eight different indicators used to assess the ability of universities to interact with the territory (environment, firms, population). In our paper we construct a synthetic index, which aggregates for each university interactions with external partners, both private firms and public institutions and we prefer it to other measures of research quality which have been used for example by D' Este and Iammarino (2010) for the UK and by Maietta (2015) for Italy.¹⁶

Importantly, the ranking of the university according to this index was completely unknown to firms since the first release was carried out in 2011 and related to the period 2004-2010. According to the Anvur assessment, universities with a higher degree of openness to external collaboration have a higher value of the overall index; we rescale the index to range between zero and one and we use this measure to weight the firm-university distance as follows:

weighted distance=distance (km)*(1+anvur_index)

We use the variable *weighted distance* to build our instrumental variable which is binary (*dummy weighted distance*) and takes value equal to 0 if the variable *weighted distance* is below the median and 1 otherwise. We consider the weighted distance between each firm and the nearest university and we compute a different median for each geographical area of the firm headquarters (North, Centre and South, defined according to the Istat classification).¹⁷ The variable *dummy weighted distance* as defined above is used as a restriction in the first stage.

¹⁵ The Anvur is the Italian National Agency for the Evaluation of University and Research Center.

¹⁶ D'Este and Iammarino (2010) find that research quality bears an influence on the frequency of collaborations between university departments and firms in the UK. Similarly, Maietta (2015) uses the quality research assessment of Italian university (VQR) to evaluate the quality of agricultural studies faculty for Italian universities.

¹⁷ The median value of the geodesic distance between the firm and the nearest university in the sample is about 17.1 km in the North, 18.8 Km in the Centre and 20.7 Km in the South.

3.6 Descriptive statistics

According to Table 1, 62.4% of firms in our sample introduce an innovation between 2008 and 2010. However, only 22.3% cooperate with universities actively as consultants or partners in a joint research program. Of the cooperating firms, 20.3% do not innovate, 4.8% introduce only organizational innovation, 21.1% introduce only technological innovation and 53.8% introduce both.

Innov	Innovation activity and the interaction with universiti								
	No innovation	Organizational	Pure Technological	Joint innovation	Total				
No Cooperation with university	372	76	151	275	874				
Cooperation with university	51	12	53	135	251				
Total	423	88	204	410	1,125				

Source: authors' calculations

Considering that we are not including small firms, these figures are substantially in line with the statistics based on the Community Innovation Survey (CIS) provided by the Italian Institute of Statistics (Istat) and confirms that our sample is not a biased representation of Italian firms.¹⁸ It is worth noting that among firms that adopt a technological innovation (both pure and joint) the share of firms that introduce a joint innovation is higher for cooperative firms.

Technological innovation

Table 2

Pure technological innovation	Joint innovation	Total
90	56	146
35	45	80
79	309	388
204	410	614
	90 35 79	90 56 35 45 79 309

Source: authors' calculations

¹⁸ According to the CIS, between 2008 and 2012 about one third of Italian firms introduced a substantial innovation in the productive process or realized an innovative good or service. However, in the same period less than 10% of interviewed firms collaborated with at least one university or other higher education institutions. In the group of firms that collaborated with universities the share of those that introduced an innovation in the productive process or a new product was much higher compared to the rest of the sample (see ISTAT 2012; data are freely available from the Istat website).

According to Table 2, in about 75% of firms with a joint innovation, the technological change involves both a new product and a new productive process; this share is much lower in pure technologically innovative firms (about 39%) which have more frequently introduced new products.

Table A3 in the appendix shows some descriptive statistics for these four groups of firms. We note that those that realized a joint innovation differ from the others, while those that introduce a pure organizational or technological innovation are on average quite similar in many respects. Broadly speaking, firms that realized a joint innovation are generally older and larger with respect to other firms in the sample. As expected, they invest more and their share of bluecollar workers over total employees is lower with respect to other firms.

The percentage of firms that export a significant share of their products is higher among technologically innovative firms, which is consistent with the results of other studies for Italy (Basile, 2001; Accetturo et al., 2014), since the propensity to export is highly correlated with the competitive attitude of firms. There is a consensus on the fact that firms which innovate are more likely to access international markets and export (see, e.g., Love and Roper, 2002; Becker and Egger, 2013). Moreover, the percentage of firms with a joint innovation is slightly higher when we look at groups of firms. Not surprisingly the variables used as proxies of firms' internal capabilities - namely the presence of a research centre (within the firm or the group) and internal R&D spending - are more widespread among these innovative firms. The share of firms that realized a technological innovation is higher in the North-West of Italy, which is also the area with higher per capita income; in contrast, the percentage of non-innovative firms is larger in the South of Italy, which is the less developed part of the country.¹⁹ When we look at Industrial districts, as defined by Istat, we find that a higher share of district firms have realized a technological or joint innovation compared to non-district firms.²⁰

Finally, firms with a technological innovation are more concentrated in the manufacturing industry, in sectors related to engineering and chemistry; in contrast, in the service sector the share of innovative firms is much lower. We think that the dichotomy is only partially related to the characteristics of the sectors. Services provide generally more labor intensive and non-tradable products, which reduce firms' incentives

¹⁹ North-West includes the following Regions: Lombardia, Piemonte, Valle d'Aosta and Liguria; North-East includes: Veneto, Trentino Alto Adige, Friuli Venezia Giulia and Emilia Romagna, Centre includes: Toscana, Lazio; Umbria, Marche; South incudes: Abruzzo, Molise, Campania, Puglia, Basilicata, Calabria, Sicilia and Sardegna.

²⁰ Istat identifies the Industrial district on the basis of "labor market areas" and the analysis of their economic specialization using the data on economic units obtained by the Industry and Services Census. In this paper we use the 2011 release by the Istat based on the 9th Industry and Services Census. The number of business clusters has decreased by 40 units with respect to the previous release (2001).

to conduct technological change. However, it also reflects a feature of the Italian economy: with respect to other advanced economies, in Italy the manufacturing industry accounts for a higher percentage of national GDP, while there is a lag in the growth of innovative services to firms (Bugamelli et al., 2018).

4. Empirical strategy

The main issue we need to address is the potential endogeneity of the variable *university*. The estimation of a non-linear model with a potentially endogenous discrete variable is a difficult task. In this paper we address this issue using the control function approach as suggested by Wooldridge (2014 and 2015). Wooldridge (2014) shows that - under certain assumptions - the control function approach allows consistent estimates of the parameters in linear models or when the endogenous variable is continuous. In contrast, the discrete nature of the endogenous variable along with non-linear models, do not lead to a simple control function estimation. However, Wooldridge (2014) claims that using the first stage residual as control functions can be - hopefully - still useful to remove most of the endogeneity in the model.

The estimated model is composed by the following two equations:

$$y_1 = g_1(y_2, x, v_1), \tag{1}$$

$$y_2 = g_2(x^+, v_2),$$
 (2)

with $x^+ = (x, z)$, where x denote the exogenous variables, while y_2 is the potentially endogenous variable and z is the instrumental variable which is correlated with the endogenous variable y_2 , but uncorrelated with the error term $v_2 : E(Z\varepsilon) = 0$. Considering equations (1) and (2) as a linear probability model (LPM), one could consistently estimates the coefficients using a two-stage least square technique. Alternatively, dealing with non-linear models we decide to adopt a control function approach. The control function approach consists in solving equation (2) for the error term and assuming the existence of a function $v_1 = h(v_2, u)$, such that u is independent of the regressors x and v_2 . Then, the control function h can be plugged into equation (1):

$$y_1 = g_1(y_2, x, v_1) = g_1(y_2, x, h(v_2, u)) = \tilde{g}_1(y_2, x, v_2, u)$$

Therefore, treating v_2 as if it were a vector of additional regressors, the error term in the model \tilde{g}_1 is u, which is independent of the regressors x and v_2 . As a result, the model \tilde{g}_1 no longer has an endogeneity problem and so can be estimated in some standard way in place of the original model g_1 .

The major limitation of this approach is that when the endogenous variable y_2 is discrete, then it is not possible that the error term v_2 is independent of x thus violating the assumption on u.²¹ However, according to Wooldridge (2014) including the first stage residuals can be still useful to deal with the endogeneity in the model.

As pointed out by Wooldridge (2015), when the potentially endogenous variable is discrete, we need to assume the way in which the control functions enter the second stage equation. The simplest approach suggested by Wooldridge (2015) is just to add the fitted error term v_2 in the second stage equation. The first to suggest plugging in residuals from a discrete first-stage estimation into a second stage were Terza et al. (2008), who called this approach 'two-stage residual inclusion'.

Turning to our model, due to the binary nature of our endogenous variable (*university*), we estimate in the first stage the following logit model:

$$P(university \neq 0 \mid x^+) = \frac{exp(x^+\beta)}{1 + exp(x^+\beta)},$$
(3)

where $x^+ = (x, weighted_distance)$, and x = (B, F, I, sectors, areas).

In equation (3) the vector of controls x^+ has been augmented by the new variable *weighted distance* with respect to the vector x. Finally, we estimate the generalized residual \hat{v}_2 from the first stage as suggested by Wooldridge (2014, 2015) and we use it as a regressor in the second stage where we estimate the following multinomial logit equation:

$$P(y_1 = m \mid x') = \frac{exp(x'\beta_m)}{1 + \sum_{h=1}^{M} x'\beta_h},$$
(4)

where $x' = (x, y_2, \hat{v}_2)$. Because we include as a regressor the residuals from the first stage estimates, we compute correct standard errors by bootstrapping equation (6) as suggested in Terza et al. (2008) and Wooldridge (2014).

Before addressing the model described by equations (3) and (4) we consider the case in which the dependent variable y_1 is binary. The dependent variable y_1 takes value equal to 1 if the firm has a technological innovation between 2008 and 2020 and zero otherwise. This choice allows us to investigate if the university cooperation can foster technological innovation regardless of whether it is a pure technological innovation or joint to an organizational change in the firm. Moreover, it allows us to address the estimation problem as a limited probability model and to gain insights on the instrumental variable.

²¹ A rigorous and detailed discussion of this approach is contained in Woodford (2014 and 2015).

5 Results on the link between Cooperation and Innovation

Accounting for firms' internal capabilities, i.e., considering the presence of absorptive capacity that enables firms to fully take advantage of the external cooperation for innovative activities, this paper aims at establishing whether there is an additional effect of firms' cooperation with universities on technological innovation. In other words, after cooperating with universities, do firms carry out technological innovations that would not have been otherwise performed?

This section summarizes our main findings; a more detailed discussion of the results is in the next two sub-sections. First, we compare no innovation activity (or only organizational) to technological innovation where we consider both 'pure' (only technological) and 'joint' (technological and organizational) innovation. Results show that cooperating with universities does not encourage additional technological innovation. This finding is partly in line with other studies which cannot find a clear-cut outcome on this aspect. What is evident in the literature is that the results depend on what one considers as the output of innovation. Some authors find an additional effect only on radical innovation (Monjon and Waelbroeck, 2003), others find an effect only on product and not on process innovation (Rouvinen, 2002) some others find the effect in the manufacturing sector but not in the service sector (Lööf and Broström, 2008)

Then, to go deeper in the analysis, we decide to investigate further the impact of cooperation on technological innovation by exploiting the dataset and splitting the effect on 'pure' and 'joint' (technological and organizational) innovation. We conduct this analysis constructing two different dependent variables and using two different sets of econometric models. In the first we consider a binary variable where we compare joint innovation to the rest, and in the second we consider a multivariate model where we consider the four different cases: no innovation,²² only organizational, only technological, and joint. These two analyses both show that, once we isolate joint innovation, cooperation becomes significant.

In particular, the multivariate model points out that there is an opposite effect on 'pure' versus 'joint' technological innovation. Firms that cooperate with universities are more likely to adopt joint innovation, whereas firms that do not cooperate with universities are more likely to adopt pure technological innovations (the coefficient on pure technological innovation is negative). This result is in line with the fact that in the bivariate model, where there is no distinction between pure and joint innovations, the impact of cooperation is not significant. In other terms, the bivariate model that encompasses all types of technological innovation

²² The category 'no innovation' includes firms that either introduced no innovation or that introduce types of innovation different from organizational and/or technological innovations.

('pure' and 'joint') in a single and broad category, cannot detect the different impact of university cooperation on the different types of technological innovations.

The fact that we find that cooperation between firms and universities only fosters those innovation activities which bind together the effort towards technological and organizational changes, is interesting since in the literature there is often a link between joint innovation and innovative performance. Ringberg et al. (2019) find for example that technological and organizational innovation together might involve a more intense change, making our results similar to those of Monjon and Waelbroeck (2003), who show an additional effect of cooperation on radical innovation. Joint innovations are relevant also because they are found to be more successful in transferring new ideas into market success (Cozzarin, 2017) and the complementarities between the two types of innovations (technological and organizational) might also increase the degree of complexity within the firm which can become an important source of competitive advantage (Rivkin, 2000).

This outcome is relevant especially for its policy implications. Policymakers should encourage the cooperation between firms and universities in Italy since the type of innovation it targets produces important changes in the economic system and turns out to be better than other types of innovation activities in promoting growth. In the next two sections we report our results and robustness checks in detail distinguishing between the two binary models and the multivariate one.

5.1 The binary model

We estimate the binary dependent variable, where 1 equals technological innovation (either pure or joint) and zero otherwise, using two different approaches. First, we estimate the binary outcomes using a linear probability model (LPM). Indeed, according to Angrist and Pischke (2009), the marginal effects of a dummy variable estimated by LPM are quite similar to those obtained by other non-linear techniques. However, addressing the endogeneity issue in linear models is much easier and can be solved by a two-stage least squares (2SLS) estimation. Moreover, in a linear model, the control function leads to the same estimated coefficients obtained by a 2SLS.

However, Lewbel et al., (2012) contrast the statement by Angrist and Pischke (2009). They claim it may occur that, using a LPM, the marginal effects may be quite different from non-linear models; moreover, in a LPM the error term cannot be independent of regressors. For this reason, we also estimate a non-linear model.

The results of the first stage are shown in Table 3. The coefficients estimated using a LPM are reported in column (a). The instrumental variable has a significant impact on the probability of cooperation and the sign is the expected one, since when the dummy variable switches from 0 to 1, the probability of cooperation between firms and university reduces of about 8 percentage points.

Table 3

		a) I	LPM	İ	b) Lo	git
	Coeffi	cient	SE	Coeffici	ient	SE
dummy weighted distance =1	-0.08111	***	0.0221	-0.0790	***	0.0216
R&D=1	0.05960	**	0.0277	0.0542	**	0.0264
research center=1	0.0984	***	0.0255	0.0834	***	0.0225
employees	0.0028		0.0063	-0.0010		0.0044
age	0.0591		0.0528	0.0389		0.0418
sales	0.0131		0.0130	-0.0022		0.0133
investments	0.0268	***	0.0071	0.0363	***	0.0098
blue-collar	-0.1675	***	0.0632	-0.2148	***	0.0682
district=1	0.0208		0.0264	0.0292		0.0259
Export						
export=1	-0.0009		0.0288	0.0018		0.0347
export=2	-0.0031		0.0400	-0.0010		0.0406
export=3	0.0592		0.0423	0.0423		0.0418
group						
group = 1	0.0210		0.0220	0.0205		0.0223
group = 2	0.0047		0.0586	-0.0042		0.0428
Obs.			1,1	25		

First stage regressions. Average marginal effects.
(Dependent variable $y_2 = 1$ if firms cooperate with the universities and 0 otherwise)

Source: authors' calculations

The coefficients are estimates of average marginal effect caused by the change in the independent variable. For factor variables discrete changes from the base case are reported. The definitions of variables are in Table A.1 in the appendix. ***, ** and * denote significance at the 1%, 5% and 10% level, respectively. Standard error, reported in *italics*, are robust to heteroskedasticity and clustering on university (80 clusters). 11 Sector and 19 Regional dummies included.

Table 4

Summary results for the first-stage regression. Tests of underidentification and weak identification

Variable	Sanderson-Windmeijer Chi-square	p-value	Kleibergen-Paap Wald rk F statistic	p-value
Cooperation	14.21	0.0002	13.52	0.0004

Source: authors' calculations.

The Kleibergen-Paap F statistic for weak identification (Table 4) shows that we can reject the null hypothesis that the instrument is weak. Similar results are also reported in the column where a logit equation is estimated.

In accordance with the literature on the complementarity between internal and external sources (Cohen and Levinthal, 1990; Becker and Dietz, 2004), the variables used to account for firm's internal capability (namely *research center* and $Re^{C}D$) turn out to affect cooperation, confirming that firms are more likely to interact for research with external partners when they are more able to benefit from the cooperation. Among other variables, the probability of firm-university cooperation increases with the size of firm's investments and when the share of blue-collars reduces.

The second stage results are reported in Table 5. The OLS estimates in column (a) shows that the cooperation with universities has a positive effect on the innovation activity. However, when we control for the possible endogeneity the marginal effect becomes statistically insignificant. Column (b) reports the results of the 2SLS regressions, which shows that university cooperation does not affect technological change (the same results are replicated in column (c) using a control function approach). On the contrary, the probability of a technological innovation is positively correlated with the firm's capability measures (the existence of a research center in the firm or group and when the dummy signaling that firm spends in R&D activities).

Column (d) contains the estimates obtained using a control function and assuming a logistic function for both equation (5) and (6). The results are similar to the 2SLS. Standard errors are bootstrapped with 200 replications. As suggested in Wooldridge (2014. 2015) the first stage residuals are 'generalized residuals' that have zero mean conditional to x^+ . Following Gourieroux et al. (1987) the generalized residuals in a logit model are defined as

$$\hat{v}_2 = y_2 - \frac{1}{1 - e^{-x^+ \hat{\beta}}},$$

where $\hat{\beta}$ are the estimated coefficients of equation (5). The first stage estimated generalized residuals are also used in the multinomial logit equation (5) whose results are discussed in the next section.

Table 5

	a) O	LS		LPM (2SLS		c) Cor	ntrol f (line	unction ar)	d)		l function ogit)
	Coefficient	SE	Coefficie	nt	SE	Coefficien	ıt	SE	Coeff	icient	SE
Cooperation	0.0644 **	0.0289	0.2180		0.3891	0.2180		0.3693	-0.0080		0.1495
generalized residuals						-0.1552		0.3652	0.0774		0.1638
R&D	0.1911 ***	0.0311	0.1820	***	0.0378	0.1820	***	0.0394	0.1908	***	0.0332
research center=1	0.1820 ***	0.0352	0.1670	***	0.0516	0.1670	***	0.0520	0.1897	***	0.0414
Employees	0.0046	0.0029	0.0041		0.0037	0.0041		0.0073	0.0093		0.0190
Age	0.0515	0.0515	0.0419		0.0508	0.0419		0.0576	0.0629		0.0554
Sales	0.0039	0.0126	0.0013		0.0127	0.0013		0.0146	0.0030		0.0143
Investments	0.0071	0.0085	0.0030		0.0133	0.0030		0.0130	0.0097		0.0100
blue-collar	-0.0158	0.0575	0.0124		0.0886	0.0124		0.0920	-0.0297		0.0700
district=1	0.0230	0.0352	0.0207		0.0374	0.0207		0.0392	0.0217		0.0349
Export											
export=1	0.1360 ***	0.0368	0.1361	***	0.0371	0.1361	***	0.0375	0.1253	***	0.0314
export=2	0.1329 ***	0.0415	0.1328	***	0.0415	0.1328	***	0.0421	0.1248	***	0.0442
export=3	0.1109 **	0.0515	0.1014	**	0.0560	0.1014	**	0.0567	0.1088	**	0.0425
Group											
group = 1	-0.0108	0.0333	-0.0146		0.0346	-0.0068		0.0312	-0.0041		0.0354
group = 2	-0.0532	0.0518	-0.0551		0.0511	-0.0648		0.0508	-0.0618		0.0555
Obs						1,125					

Second stage regressions. Average marginal effects. (Binary dependent variable y1 =1 if firms introduced a technological innovation and 0 otherwise)

Source: authors' calculations.

The coefficients are estimates of average marginal effect caused by the change in the independent variable. For factor variables discrete changes from the base case are reported. The definitions of variables are in Table A.3. ***, ** and * denote significance at the 1%, 5% and 10% level, respectively. Standard errors are reported in *italics*: in columns a) and b) standard errors are robust to heteroskedasticity and clustering on university (80 clusters); in columns c) end d) standard error are bootstrapped with 200 replications. 11 Sector and 19 Regional dummies included.

In order to gain more insights about the interaction between cooperation and innovation, we adopt a different definition of the binary dependent variable y_1 , which takes value equal to 1 when firms have introduced a joint innovation and zero otherwise. Once again, the second stage results are obtained assuming alternatively a LPM or logistic. The results are displayed in Table 6 (columns b-d). We find that the university cooperation is statistically significant in fostering firms' joint innovation activity. More precisely, the cooperation with a university increases the probability of joint innovation of about 87 percentage points when we use a two-stage estimation of LPM. However, the marginal effects reduce to about 34 percentage points when we use a control function where both the first and the second stage are modelled as logit equations.

Among other controls, we still find that the joint innovation activity is positively affected by the presence of a research center and R&D expenditure.

		a) OLS		b)	b) LPM c) Control func (linear)				tion d) Control function (logit)			
	Coeffic	eint	SE	Coefficie	nt	SE	Coefficie	ent	SE	Coefficie	ent	SE
Cooperation	0.0512		0.0363	0.8729	**	0.4057	0.8729	**	0.3784	0.3435	**	0.1518
generalized residuals							-0.8300	**	0.3789	-0.2993	*	0.1811
R&D=1	0.1086	***	0.0307	0.0601		0.0456	0.0601		0.0382	0.0826	**	0.034
research center=1	0.1239	***	0.0357	0.0435		0.0588	0.0435		0.0522	0.0823	**	0.0380
Employees	0.0040		0.0034	0.0014		0.0066	0.0014		0.0091	0.0057		0.0180
Age	0.0038		0.0552	-0.0476		0.0693	-0.0476		0.0609	-0.0135		0.0650
sales	0.0249	**	0.0113	0.0109		0.0151	0.0109		0.0143	0.0139		0.0139
investments	0.0134	*	0.0079	-0.0083		0.0125	-0.0083		0.0120	0.0079		0.009-
blue-collar	-0.0313		0.0527	0.1195		0.0944	0.1195		0.0876	-0.0020		0.0746
district=1	0.0146		0.0344	0.0021		0.0384	0.0021		0.0348	0.0084		0.0357
Export												
export=1	0.0745	*	0.0398	0.0749		0.0486	0.0749	*	0.0430	0.0812	**	0.0365
export=2	0.0744	*	0.0446	0.0738		0.0561	0.0738		0.0456	0.0723		0.0471
export=3	0.0744		0.0499	0.0236		0.0698	0.0236		0.0558	0.0523		0.0468
Group												
group = 1	0.0105		0.0308	-0.0098		0.0356	-0.0098		0.0275	0.0049		0.0347
group = 2	-0.0195		0.0528	-0.0298		0.0676	-0.0298		0.0471	-0.0249		0.049

Second stage regressions. Average marginal effects. (Binary dependent variable y1 =1 if firms introduced a joint innovation and 0 otherwise)

Table 6

Source: authors' calculations.

The coefficients are estimates of average marginal effect caused by the change in the independent variable. For factor variables discrete changes from the base case are reported. The definitions of variables are in Table A.3. ***, ** and * denote significance at the 1%, 5% and 10% level, respectively. Standard errors are reported in *italics*: in columns a) and b) standard errors are robust to heteroskedasticity and clustering on university (80 clusters); in columns c) end d) standard error are bootstrapped with 200 replications. 11 Sector and 19 Regional dummies included.

5.2 The multivariate model

Tables 7 and 8 report the average marginal effects of the second stage of the multinomial logit, which describe how the control variables affect the probability of realization for each considered outcome. In the model the dependent variable has four mutually exclusive outcomes (*no innovation, organizational, technological, joint innovation*) and where the estimated residuals from the first stage (table 3) have been included among the regressors.²³ Bootstrapped standard errors are obtained by 200 replications. In table 7 the first stage is a LPM, while in table 8 we used a non-linear model. All the main results are similar.

The main finding reported in the tables is that the interaction with universities increases the probability of a joint innovation strategy by the firm but reduces the likelihood of a pure technological innovation with respect to non-collaborating firms. The results are interesting and shed some lights on the binary model outcome, where we have not distinguished between the firms' innovation strategies. Although the cooperation with universities does not increase the probability of technological innovation of firms, it differently targets the various types of technological innovation. In other terms universities stimulate firms to introduce new products or productive processes together with some organizational change. In contrast, it is less likely to observe firms that cooperate with universities and just adopt new products or productive processes that do not require any organizational or managerial change. Then, the multinomial model results are consistent with the findings displayed in table 8 where the binary dependent variable is equal to 1 in case of joint innovation and zero otherwise.

Finally, it is worth noticing that also in the multivariate model the measures of firms' capabilities (*research* and $R \notin D$) have statistically significant effects on the probability of both pure technological and joint innovations.

²³ The outcome variable is constructed so that each firm can not be in multiple categories. The estimated multinomial logit model satisfies the property of the independence of irrelevant alternatives (IIA), according to which the odds ratios should not be affected by the availability of alternative choices. Indeed, the introduction of an extra type of innovation does not change the firms' choice on previous innovation types.

	No	innova	tion	Orgai	nizat	tional	Tecl	molog	gical		Joint	
	Coeffici	ient	SE	Coefficie	ent	SE	Coeffic	ient	SE	Coeffic	eint	SE
cooperation	-0.1775		0.1766	-0.1601	*	0.0862	-0.2033	**	0.0905	0.5409	***	0.1729
generalized residuals	-0.2128		0.3794	0.3907	*	0.2287	0.6846	***	0.3363	-0.8625	***	0.3322
R&D=1	-0.1658	***	0.0393	-0.0127		0.0220	0.1307	***	0.0332	0.0478		0.0373
research center=1	-0.1541	***	0.0521	-0.0099		0.0321	0.1345	***	0.0453	0.0295		0.0468
employees	-0.0003		0.0295	-0.0057		0.0239	0.0036		0.0566	0.0024		0.0357
Age	-0.0464		0.0616	-0.0041		0.0353	0.0895	*	0.0520	-0.0390		0.0655
Sales	-0.0157		0.0169	0.0176	*	0.0104	-0.0094		0.0144	0.0074		0.0171
investments	-0.0137		0.0132	0.0103		0.0090	0.0086		0.0127	-0.0053		0.0127
blue-collar	0.0657		0.0967	-0.0817		0.0531	-0.1068		0.0898	0.1228		0.0910
district=1	-0.02217		0.0357	0.0051		0.0247	0.0126		0.0333	0.0045		0.0335
export												
export=1	-0.1322	***	0.0397	0.0079		0.0227	0.0507		0.0313	0.0735	*	0.0379
export=2	-0.0945	*	0.0508	-0.0313		0.0280	0.0568		0.0367	0.0690		0.0459
export=3	-0.0877		0.0542	-0.0101		0.0367	0.0849	*	0.0497	0.0128		0.0524
group												
group = 1	0.0117		0.0353	-0.0020		0.0194	-0.0003		0.0280	-0.0094		0.0299
group = 2	0.0384		0.0599	0.0311		0.0429	-0.0240		0.0441	-0.0454		0.0483
Obs.						1,1	25					

Second stage Multinomial Logit Regression. Average marginal effects. (First stage: linear probability model)

Source: authors' calculations.

The coefficients are estimates of average marginal effect caused by the change in the independent variable. For factor variables discrete changes from the base case are reported. The definitions of variables are in Table A.3. ***, ** and * denote significance at the 1%, 5% and 10% level, respectively. Bootstrapped Standard errors (200 replications) are reported in *italics*. Sector and Regional dummies included.

Table 8

Second stage Multinomial Logit Regression. Average marginal effects. *(First stage: logit model)*

	No inn	ovation	Organiza	tional	Tec	hnolog	rical		Joint	
	Coefficient	SE	Coefficient	SE	Coeffic	C	SE	Coeffici	•	SE
cooperation	0.0505	0.1538	-0.0862	0.0628	-0.2354	***	0.0682	0.2710	*	0.1535
generalized residuals	-0.2652	0.2011	0.1736	0.1242	0.4199	**	0.1730	-0.3283	*	0.1818
R&D=1	-0.1682 ***	* 0.0347	-0.0264	0.0185	0.1141	***	0.0287	0.0806	**	0.0323
research center=1	-0.1591 **:	* 0.0381	-0.0230	0.0223	0.1041	***	0.0315	0.0870	**	0.0391
employees	-0.0017	0.0279	-0.0072	0.0199	0.0046		0.0434	0.0043		0.0292
age	-0.0457	0.0575	-0.0184	0.0370	0.0694		0.0478	-0.0053		0.0590
sales	-0.0145	0.0149	0.0138	0.0098	-0.0157		0.0131	0.0164		0.0149
investments	-0.0141	0.0112	0.0036	0.0071	0.0035		0.0111	0.0070		0.0124
blue-collar	0.0707	0.0749	-0.0319	0.0384	-0.0830		0.0797	0.0442		0.0864
district=1	-0.0226	0.0393	0.0001	0.0235	0.0117		0.0326	0.0109		0.0376
export										
export=1	-0.1333 ***	* 0.0366	0.0092	0.0232	0.0521		0.0335	0.0720	*	0.0377
export=2	-0.0964 **	0.0448	-0.0292	0.0246	0.0549		0.0399	0.0708		0.0459
export=3	-0.0915 *	0.0471	-0.0188	0.0285	0.0618		0.0461	0.0484		0.0453
group										
group = 1	0.0127	0.0339	-0.0106	0.0200	-0.0035		0.0298	0.0014		0.0346
group = 2	0.0361	0.0573	0.0267	0.0406	-0.0241		0.0455	-0.0387		0.0486
Obs				1,1	25					

Source: authors' calculations.

The coefficients are estimates of average marginal effect caused by the change in the independent variable. For factor variables discrete changes from the base case are reported. The definitions of variables are in Table A.3. ***, ** and * denote significance at the 1%, 5% and 10% level, respectively. Bootstrapped Standard errors (200 replications) are reported in *italics*. Sector and Regional dummies included.

6 Concluding remarks

We have provided robust evidence about the effects of the university cooperation on the technological innovation activity of Italian firms. The cooperation with universities does not encourage additional technological innovation, whereas it increases the likelihood of firms adopting a joint innovation strategy, which encompasses both technological and organizational improvements. The innovation management literature points out that the latter kind of innovation is the one that firms should preferably pursue to achieve market success. The results, based on data collected in the past decade, suggest that the cooperation with universities may be particularly useful for Italian firms to reduce the technological gap and increase productivity with respect to their main competitors. However, this empirical evidence is at odds with the low percentage of Italian firms collaborating with one or more universities, which is confirmed also by the most recent available data based on CIS 2018. This percentage is lower with respect to other advanced economies and imply that further efforts by the policymaker are needed to stimulate the firm-university cooperation.

Appendix

	T		•	regressions
	I tet ot	vamablee	in the	#00#00010m0
I ADIC A.I.	LISCUL	valiabics	III UIC	102103310113

Variable name	Description	Nature
Dependent variables		
Innovation	Innovative outcome	Factor variable
innovation=0	No innovation	
innovation=1	Only managerial or organizational innovation	
innovation=2	New productive process and/ or new product	
innovation=3	Joint innovation (technological and non-technological)	
Instrument	, , , , , , , , , , , , , , , , , , , ,	Dummy
Dummy weighted	If distance (in Kms) between firms and universities,	
distance=0	weighted for the Anvur index about the third mission below the median value	
Dummy weighted distance=1	If distance (in Kms) between firms and universities, weighted for the Anvur index about the third mission	
Independent variables	above the median value	
University	Type of collaboration with university	Dummy
university=0	No collaboration or Firm hosting a Ph.d students / internships	
university=1	Consultancy and/or partnership in a research project	
Group	Type of collaboration with university	Factor variable
group=0	Firm does not belong to a group	
group=1	Firm belongs to an Italian group	
group=2	Firm belongs to a foreign group	
Age	Average age (100 years)	Continuous
employees	Average number of Employees (1,000 units)	Continuous
sales	Average sales (in natural logarithms)	Continuous
investments	Average investments (in natural logarithms)	Continuous
blue-collar	Average share of blue-collar workers	Continuous
District	Firm located in an industrial district	(between 0 and 1) Dummy
district=0	Firm is not located in an industrial district	Dunniy
district=1	Firm is located in an industrial district	
Export	Share of sales from export	Factor variable
export=0	Non-exporting firm	
export=1	Export less than 1/3 of total sales	
export=2	Export between 1/3 and 2/3 of total sales	
export=3	Export more than $2/3$ of total sales	
R&D	Internal R&D spending o firm	Dummy
R&D=0	Firm does not spend in internal R&D	
R&D=1	Positive firm's spending in R&D	D
Research center		Dummy
research center=0	No research center in Italy or abroad	
research center=1	Firm has a research center in Italy or abroad	

Source: authors' calculations.

We do not report in the table 11 sectoral dummies and 18 regional dummies.

	Coop	R&D	Research center	Employ- ment	Age	Invest- ments	Sales	Blue- collar	Export	District	Group
Соор	1										
R&D	0.2025	1									
Research center	0.2716	0.2745	1								
Employment	0.1251	0.1098	0.1132	1							
Age	0.1016	0.1156	0.1014	0.0609	1						
Investments	0.2896	0.2751	0.2418	0.3492	0.1617	1					
Sales	0.2765	0.2990	0.2622	0.3886	0.1900	0.7408	1				
Blue-collars	-0.1281	-0.1311	-0.1255	-0.1131	-0.0530	-0.1299	-0.3226	1			
Export	0.1733	0.2313	0.3092	0.0263	0.1258	0.1280	0.1575	0.0477	1		
District	0.0051	0.1193	0.0653	-0.0414	0.0400	-0.0814	-0.0148	0.0289	0.1609	1	
Group	0.2038	0.1952	0.3685	0.1321	0.0567	0.3879	0.4616	-0.2351	0.2108	-0.0561	1

Table A2. Correlation matrix of the regressors.

Source: authors' calculations.

Variables	No innovation		Pure Organizational innovation		Pure technological innovation		Joint Innovation		Full sample	
	Mean	Median	Mean	Median	Mean	Median	Mean	Median	Mean	Median
Age (years)	32.9	28	33.1	28	38.9	35	39.7	34	36.5	31
Number of Employees (units)	215	61	302	72	230	57	741	120	416	76
Sales (millions of euro)	58	12	172	19	209	11	246	26	163	17
Investments (millions of euro)	7.5	0.6	21.2	1.0	14.0	0.7	31.7	1.6	18.6	0.9
Share of blue-collar workers (percentage points)	63.7	73.3	57.3	63.5	66.6	70.2	61.2	67.4	62.8	70.0
					(per	centage)				
University collaboration					F				,	
No University collaboration	74.5		67.1		59.3		53.9		62.7	
Loose collaboration (Internship)	13.5		19.3		14.7		15.5		15.0	
Collaboration (consultancy, partnership) Research center	12.0 21.5		13.6 15.9		26.0 49.5		30.6 57.8		22.3 39.4	
Internal R&D spending	21.5 29.1		31.8		49.5 60.3		66.1		48.4	
Share of exporting sales:		27.1		51.0		00.5		00.1	T T	0.7
Zero		47.3	F	50.0	,	19.1		16.3	3	1.1
Less than 1/3	29.8		36.4		41.2		39.0		35.7	
Between $1/3$ and $2/3$	12.0		6.8		23.5		25.4		18.6	
More than $2/3$		10.9		6.8		16.2		19.3	14	4.6
Firms' groups										
No group		64.5		50.2		50.8		45.4		6.5
Italian group	29.8		31.8		31.9		43.4		35.3	
Foreign group	5.7		8.0		7.3		11.2		8.2	
Belong to an Industrial District	20.6		20.5		31.4		32.0		26.7	
Geographical location:										
North-West	18.9		17.1		26.5		29.3		23.9	
North-East	11.6		20.4		15.2		21.5		16.5	
Centre	21.5		30.7		24.0		26.1		24.4	
South and Isle	48.0		31.8		34.3		23.2		35.2	
Economic sector:										
Food and Beverage		12.1		4.6	,	14.2		12.7	1	2.1
Textile, clothing, leather, shoes	6.6		2.3		13.7		10.7		9.1	
Chemicals, rubber and plastics	4.7		5.7		9.8		11.2		9.1 8.1	
Energy and Extraction	4.7 5.7		5.7 8.0		9.8 8.3		6.1			
Engineering									6.5	
0 0	23.6		13.6		34.3		36.6		29.5	
Other manufacturing	6.1		1.1		10.8		9.5		7.8	
Other not-manufacturing	3.8		6.8		3.9		0.2		2.7	
Wholesale and retail trade	18.2		22.7		2.5		5.4		11.0	
Hotels and restaurants	2.4		9.1		0		1.0		2.0	
Transport and communication	11.6		19.3		1.5		3.9		7.6	
Other business and household services	5.2		6.8		1.0		2.7		3.6	

Table A3 Descriptive statistics according to innovative activity of firms.

Source: authors' calculations.

References

- Accetturo, A., Bugamelli M., Lamorgese A.R. and Linarello A. (2014), Innovation and trade. Evidence from Italian manufacturing firms, mimeo posted, Rome, Italy.
- Acs, Z. J., Braunerhjelm, P., Audretsch, D. B. and Carlsson, B. (2009). The knowledge spillover theory of entrepreneurship. *Small business economics*, 32, 15-30.
- Aksoy, A. Y., Pulizzotto, D., and Beaudry, C. (2022). University-Industry partnerships in the smart specialisation era. *Technological Forecasting and Social Change*, 176, 121438.
- Amesse, F. and Cohendet P. (2001), Technology Transfer Revisited from the Perspective of the Knowledge-Based Economy, *Research Policy*, 30, 1459-1478.
- Angrist, J.D. and Pischke, J.S. (2009), Mostly Harmless Econometrics: An Empiricists Companion, Princeton University Press, Princeton, NJ.
- Arvanitis, S., Seliger, F. and Woerter, M. (2020), Knowledge Spillovers, Competition and Innovation Success, Oxford Bulletin of Economics and Statistics, 82, 1017-1041.
- Arvanitis, S., Sydow, N. and Woerter, M. (2008), Is there any impact of university–industry knowledge transfer on innovation and productivity? An empirical analysis based on Swiss firm data, Review of Industrial Organization, 32, 77-94.
- Arrow, K. J. (1962), The economic implications of learning by doing. The Review of Economic Studies, 29, 155– 173.
- Aschhoff, B. and Schmidt (2008), Empirical Evidence on the Success of R&D Cooperation Happy Together?, *Review of Industrial Organization*, 33, 41-62.
- Audretsch, D. B. (2014). From the entrepreneurial university to the university for the entrepreneurial society. *The Journal of Technology Transfer*, 39, 313-321.
- Audretsch, D. B., Keilbach, M. C. and Lehmann, E. E. (2006). Entrepreneurship and economic growth. Oxford University Press.
- Audretsch, D. B. and Feldman, M. P. (1996), R&D spillovers and the geography of innovation and production, *The American economic review*, 86, 630-640.

- Ballot, G., Fakhfakh, F., Galia, F. and Slater, A. (2015), The fateful triangle: Complementarities in performance between products, process and organizational innovation in France and the UK, *Research Policy*, 44, 217-232.
- Baranano, A.M. (2003), The non-technological side of technological innovation: state-of-the-art and guidelines for further empirical research, *International Journal of Entrepreneurship and Innovation Management*, 3, 107-125.
- Barbieri, L., Bragoli, D., Cortelezzi, F. and Marseguerra, G. (2020), Public Funding and Innovation Strategies. Evidence from Italian SMEs, *International Journal of the Economics of Business*, 27: 111-134.
- Bartoloni, E. and Baussola M. (2018), Driving business performance: innovation complementarities and persistence patterns, *Industry and Innovation*, 25, 505-525.
- Basile, R. (2001), Export Behaviour of Italian Manufacturing Firms Over the Nineties: The Role of Innovation, Research Policy, 30, 1185-1201.
- Battisti, G. and Iona, A. (2009), The intra-firm diffusion of complementary innovations: Evidence from the adoption of management practices by British establishments, *Research Policy*, 38, 1326-1339.
- Becker, W. (2003), Evaluation of the Role of Universities in the Innovation Process, Discussion Paper Series No. 241, Universitaet Augsburg, Institute for Economics, Augsburg, Germany.
- Becker, S. O. and Egger, P. H. (2013), Endogenous product versus process innovation and a firms propensity to export, *Empirical Economics*, 44, 329-354.
- Becker, W. and Dietz, J. (2004), R&D cooperation and innovation activities of firm evidence for the German manufacturing industry, *Research policy*, 33, 209-223.
- Belderbos, R., Carree, M. and Lokshin, B. (2004), Cooperative R&D and firm performance, *Research policy*, 33, 1477-1492.
- Bellandi, M., Caloffi, A. and De Masi, S. (2021). Bottom-level organizational changes within entrepreneurial and engaged models of university: insights from Italy. *The Journal of Technology Transfer*, 46, 907-932.
- Berchicci, L. (2013), Towards an open R&D system: Internal R&D investment, external knowledge acquisition and innovative performance, *Research policy*, 42, 117-127.
- Boer, H. and During, W.E. (2001), Innovation, what innovation? A comparison between product, process and organizational innovation, *International Journal of Technology Management*, 22, 83-107.
- Birkinshaw, J., Hamel, G. and Mol, M. (2008), Management innovation, *Academy of Management Review*, 33, 825-845.
- Bragoli, D., Cortelezzi, F. and Marseguerra, G. (2016), R&D, capital structure and ownership concentration: evidence from Italian microdata, *Industry and Innovation*, 23, 223-242.

- Bragoli, D., Cortelezzi F., Marseguerra, G. and Rigon, M. (2020), Innovative investments, financial imperfections and the Italian business cycle, *Oxford Economic Papers*, 72, 412-434.
- Branstetter, L. and Ogura, Y. (2005), Is academic science driving a surge in industrial innovation? Evidence from patent citations, National Bureau of Economic Research, n. 11561.
- Bugamelli, M., Cannari, L., Lotti, F. and Magri, S. (2012), The Innovation Gap of Italys Production System: Roots and Possible Solutions, *Occasional Papers*, 121, Bank of Italy.
- Bugamelli M., Lotti, F., Amici, M., Ciapanna, E., Colonna, F., D'Amuri, F., Giacomelli, S., Linarello, A., Manaresi, F., Palumbo, G., Scoccianti, F. and Sette, E. (2018), Productivity growth in Italy: a tale of a slow-motion change, *Occasional Papers*, 422, Bank of Italy.
- Caloghirou, Y., Kastelli, I. and Tsakanikas, A. (2004), Internal capabilities and external knowledge sources: complements or substitutes for innovative performance?, *Technovation*, 24, 29-39.
- Caloghirou, Y., Giotopoulos, I., Kontolaimou, A., Korra, E. and Tsakanikas, A. (2021). Industry-university knowledge flows and product innovation: How do knowledge stocks and crisis matter?, *Research Policy*, 50, 104-195.
- Cassiman, B. and Veugelers, R. (2002), R&D cooperation and spillovers: some empirical evidence from Belgium, *American Economic Review*, 92, 1169-1184.
- Cassiman, B. and Veugelers, R. (2005), R&D cooperation between firms and universities. Some empirical evidence from Belgian manufacturing, *International Journal of Industrial Organization*, 23, 355-379.
- Cerne, M., Kaše, R. and Škerlavaj, M. (2016), Non-technological innovation research: Evaluating the intellectual structure and prospects of an emerging field, *Scandinavian Journal of Management*, *32*, 69-85.
- Cerver Romero, E., Ferreira, J.J.M. and Fernandes, C.I. (2021) The multiple faces of the entrepreneurial university: a review of the prevailing theoretical approaches, *The Journal of Technological Transfer*, 46, 1173–1195.
- Chesbrough, H.W. (2003), Open Innovation: The New Imperative for Creating and Profiting from Technology, *Harvard Business School Press*, Boston, MA.
- Cohen, W.M. and Levinthal, D.A. (1990), Absorptive capacity. A new perspective on learning and innovation, *Administrative Science Quarterly*, 35, 128-152.
- Cozzarin, B.P. (2017), Impact of organizational innovation on product and process innovation, *Economics of Innovation and New Technology*, 26, 405-417.
- Cozzarin, B.P. and Percival, J.C. (2006), Complementarities between organizational strategies and innovation, *Economics of Innovation and New Technology*, 15, 195-217.

- Cozzarin, B.P. and Percival, J.C. (2008), Complementarities Affecting the Returns to Innovation, *Industry and Innovation*, 15, 371-392.
- Crossan, M. M. and Apaydin, M. (2010), A multi-dimensional framework of organizational innovation: A systematic review of the literature, *Journal of Management Studies*, 47, 1154-1191.
- Daft, R. L. (1978), A dual-core model of organizational innovation, *Academy of Management Journal*, 21, 193-210.
- Damanpour, F. and Evan, W. M. (1984), Organizational innovation and performance: The problem of organizational lag, *Administrative Science Quarterly*, 29, 392-409.
- Damanpour, F., Walker, R. M. and Avellaneda, C. N. (2009), Combinative effects of innovation types and organizational performance: A longitudinal study of services organizations, *Journal of Management Studies*, 46, 650–675.
- Damanpour, F. (2020), Organizational innovation: Theory, research, and direction. Edward Elgar Publishing, Cheltenham, UK.
- De Bondt, R. (1996), Spillovers and Innovative activities, *International Journal of Industrial Organization*, 15, 1-28.
- D'Este, P., and Iammarino, S. (2010), The spatial profile of university-business research partnerships, *Papers in regional science*, 89, 335-350.
- D'Este, P., Guy, F., and Iammarino, S. (2013), Shaping the formation of university-industry research collaborations: what type of proximity does really matter?, *Journal of Economic Geography*, 13, 537-558.
- Drejer, A. (2002), Situations for innovation management: towards a contingency model, *European Journal of Innovation Management*, 5, 4-17.
- European Commission (1995), Green Paper on Innovation, European Commission, Bruxelles.
- European Commission (2017) Annual Report on SMEs 2016/2017, European Commission, Bruxelles.
- Evangelista, R. and Vezzani, A. (2010), The economic impact of technological and organizational innovations. A firm-level analysis, *Research Policy*, 39, 1253-1263.
- Etzkowitz, H. and Leydesdorff, L. (1997), Introduction to special issue on science policy dimensions of the Triple Helix of university-industry-government relations, *Science and Public Policy*, 24, 2-5.
- Fantino, D., Mori, A. and Scalise, D. (2015), Collaboration between firms and universities in Italy: the role of a firms proximity to top-rated departments, *Italian Economic Journal*, 1, 219-251.
- Fritsch, M., and Franke, G. (2004), Innovation, regional knowledge spillovers and R&D cooperation, Research policy, 33, 245-255.

- Garcia, R. and Calantone, R. (2002), A critical look at technological innovation typology and innovativeness terminology: a literature review, *Journal of Product Innovation Management: An international publication of the product development & management association*, 19, 110-132.
- Georgantzas, N. C., and Shapiro, J. H. (1993), Viable theoretical forms of synchronous product innovation, Journal of Operations Management, 11, 161-183.
- Giuliani, E. and Arza, V. (2009), What drives the formation of valuable university-industry linkages? Insights from the wine industry, *Research Policy*, 38, 906-921.
- Gómez, J., Salazar, I. and Vargas, P. (2020), The role of extramural R&D and scientific knowledge in creating high novelty innovations: an examination of manufacturing and service firms in Spain, *Research Policy*, 49, article 104030.
- Gourieroux, C., Monfort, A., Renault, E. and Trognon, A. (1987), Generalised residuals, *Journal of Econometrics*, 34, 5-32.
- Grimaldi, R., Kenney, M. and Piccaluga, A. (2021). University technology transfer, regional specialization and local dynamics: lessons from Italy. *The Journal of Technology Transfer*, 46(4), 855-865.
- Hagedoorn, J., and Wang, N. (2012), Is there complementarity or substitutability between internal and external R&D strategies?, *Research policy*, 41, 1072-1083.
- Hall, B. H. (2004). University-industry research partnerships in the United States, *EUI Working Paper ECO*, No. 2004 /14.

Hall, B. H., Link, A.N. and Scott, J. T. (2001), Barriers Inhibiting Industry from Partnering with Universities. *Journal of Technology Transfer*, 26, 87-98.

- Hall, B. H., Link, A.N. and Scott, J. T. (2003), Universities as Research Partners, *Review of Economics and Statistics*, 85, 485-491.
- Hall, B. H., Lotti F. and Mairesse J. (2013) Evidence on the impact of R&D and ICT investments on innovation and productivity in Italian firms, *Economics of Innovation and New Technology*, 22, 300-328.
- Hamel, G. (2006), The why, what and how of management innovation, Harvard Business Review, 84, 72-84.
- Harabi, N. (1995), Channels of R&D spillovers: an empirical investigation, *WWI-Arbeitspapiere*, Reihe D No. 37, 1-20.
- Hervas-Oliver, J. L., Sempere-Ripoll, F., & Boronat-Moll, C. (2012), Process innovation objectives and management complementarities: patterns, drivers, co-adoption and performance effects, MERIT Working papers No. 51, United Nations University. Maastricht, Netherlands.
- Hervas-Oliver, J.-L. and Sempere-Ripoll, F. (2014), Disentangling the influence of technological process and product innovations, *Journal of Business Research*, 68, 109-118.

- Hewitt-Dundas, N. (2013), The role of proximity in university-business cooperation for innovation, *The Journal of Technology Transfer*, 38, 93–115.
- Hipp, C. and Grupp, H. (2005), Innovation in the Service Sector: The Demand of Service-specific Innovation Measurement Concepts and Typology, *Research Policy*, 34, 517-535.
- Istat (2012), Innovazione delle imprese, Istat, Rome, Italy
- Jaffe, A. B. (1989), Real effects of academic research, The American Economic Review, 79, 957-970.
- Johannessen, J. A., Olsen, B. and Lumpkin, G. T. (2001), Innovation as newness: what is new, how new, and new to whom? *European Journal of Innovation Management*. 4, 20-31.
- Kochenkova, A., Grimaldi, R. and Munari, F. (2016), Public policy measures in support of knowledge transfer activities: a review of academic literature, *Journal of Technology Transfer*, 41, 407-429.
- Landry, R., Amara, N. and Ouimet, M. (2007), Determinants of knowledge transfer: evidence from Canadian university researchers in natural sciences and engineering, *The Journal of Technology Transfer*, 32, 561-592.
- Laursen K., Reichstein, T. and Salter, A. (2008), Exploring the Effects of Geographical Proximity on Industry-University collaboration in the UK, *Regional Studies*, 45, 507-523.
- Laursen, K. and Salter, A. (2006), Open for innovation: the role of openness in explaining innovation performance among UK manufacturing firms, *Strategic Management Journal*, 27, 131-150.
- Lee, Y. S. (2000), The Sustainability of University-Industry Research Collaboration, *The Journal of Technology Transfer*, 25, 111-133.
- Lewbel, A., Dong, Y. and Yang, T.T. (2012), Comparing features of convenient estimators for binary choice models with endogenous regressors, Canadian Journal of Economics, 45, 809-829.
- Link, A. N. and Sarala, R. M. (2019). Advancing conceptualisation of university entrepreneurial ecosystems: The role of knowledge-intensive entrepreneurial firms. *International Small Business Journal*, 37, 289-310.
- Lööf, H. and Broström, A. (2008), Does knowledge diffusion between university and industry increase innovativeness? *The Journal of Technology Transfer*, 33, 73-90.
- Love, J. H. and Roper, S. (2002), Innovation and export performance: evidence from the UK and German manufacturing plants, *Research Policy*, 31, 1087-1102.
- Maietta, O. W. (2015), Determinants of university-firm R&D collaboration and its impact on innovation: A perspective from a low-tech industry, *Research Policy*, 44, 1341-1359.
- Mansfield, E. (1991), Academic Research and Industrial Innovation, Research Policy 20, 1-12.

- Mansfield, E. (1995), Academic research underlying industrial innovations: sources, characteristics, and financing, *Review of Economics and Statistics*, 77, 55-65.
- Mansfield, E. and Lee, J.Y. (1996), The modern university: contributor to industrial innovation and recipient of industrial R&D support, *Research Policy*, 25, 1047-1058.
- Marseguerra, G., Bragoli, D. and Cortelezzi, F. (2016), Assessing the innovative performance of Italian SMEs, *Istituto Lombardo-Accademia di Scienze e Lettere-Rendiconti di Lettere*, Milan, Italy.
- Meeus, M. T. H. and Edquist, C. (2006), Introduction to Part I: Product and process innovation, in J. Hage and M. Meeus (Eds.), *Innovation, science, and institutional change.* Oxford University Press, 23–37.
- Medda, G., Piga, C. and Siegel, D. S. (2006), Assessing the returns to collaborative research: Firm-level evidence from Italy. *Economics of Innovation and New technology*, 15, 37-50.
- Monjon, S. and P. Waelbroeck (2003), Assessing Spillovers from Universities to Firms: Evidence from French Firm-Level Data, *International Journal of Industrial Organization*, 21, 1255-1270.
- Morgan, K. (2004), The exaggerated death of geography: learning, proximity and territorial innovation systems, *Journal of Economic Geography*, 4, 3-21.
- Mowery, D. and Sampat, N. (2005), Universities in national innovation systems, in J. Fargerberg, D. Mowery and R. Nelson (Eds.), *The Oxford Handbook of Innovation*, Oxford University Press, Oxford.
- Nickell, S., Nicolitsas, D. and Patterson, M. (2001), Does doing badly encourage management innovation?, Oxford Bulletin of Economics and Statistics, 63, 5-28.
- Naranjo-Gil, D. (2009), The influence of environmental and organizational factors on innovation adoption: Consequences for performance in public sector organizations. *Technovation*, 29, 810-818.
- OECD (2018), Science, Technology and Innovation Outlook, OECD Publishing, Paris, France.
- OECD and Eurostat (2005), Oslo Manual Proposed Guidelines for Collecting and Interpreting Technological Innovation Data - 3rd Edition, Paris, France
- Parmentola, A., Ferretti, M. and Panetti, E. (2021). Exploring the university-industry cooperation in a low innovative region. What differences between low tech and high tech industries? *International Entrepreneurship and Management Journal*, 17(3), 1469-1496.
- Pellegrino, G., Piva, M. and Vivarelli, M. (2012), Young firms and innovation: A microeconometric analysis *Structural Change and Economic Dynamics*, Elsevier, 23, 329-340.
- Ringberg, T., Reihlen, M. and Rydén, P. (2019), The technology-mindset interactions: Leading to incremental, radical or revolutionary innovations, *Industrial Marketing Management*, 79, 102-113.

- Rippa, P. and Secundo, G. (2019). Digital academic entrepreneurship: The potential of digital technologies on academic entrepreneurship. *Technological Forecasting and Social Change*, 146, 900-911.
- Rivkin, J.V. (2000), Imitation of complex strategies, Management Science, 46, 824-844.
- Roberts, P. W. and Amit, R. (2003), The dynamics of innovative activity and competitive advantage: The case of Australian retail banking, 1981 to 1995. *Organization Science*, 14, 107–122.
- Robin, S. and Schubert, T. (2013), Cooperation with public research institutions and success in innovation: Evidence from France and Germany, *Research Policy*, 42, 149-166.
- Rouvinen, P. (2002), Characteristics of product and process innovators: some evidence from the Finnish innovation survey, *Applied Economic Letters*, 9, 575-580.
- Sanidas, E. (2005), Organizational innovations and economic growth: Organosis and growth of firms, sectors, and countries. Edward Elgar Publishing Limited, Cheltenham, U.K.
- Scandura, A. (2016), University-industry collaboration and firms R&D effort, Research Policy, 45, 1907-1922.
- Schmidt, T. and Rammer, C. (2007), Non-technological and technological innovation: strange bedfellows?, ZEW-Centre for European Economic Research Discussion Paper, 07-052, Mannheim, Germany.
- Secundo, G., Mele, G., Del Vecchio, P. and Degennaro, G. (2021). Knowledge spillover creation in university-based entrepreneurial ecosystem: the role of the Italian Contamination Labs. *Knowledge Management Research & Practice*, 19(1), 137-151.
- Sforzi, F. (2009), The empirical evidence of Industrial Districts in Italy in G. Becattini, M. Bellandi and L. De Propis (eds.), *A Handbook of Industrial Districts*, 327-341, Edward Elgar Publishing Limited, New York.
- Spence, M. (1984), Cost reduction, competition, and industry performance. *Econometrica: Journal of the Econometric Society*, 52, 101-122.
- Terza J.V., Basu, A. and Rathouz, P.J. (2008), Two-stage residual inclusion estimation: Addressing endogeneity in health econometrics modeling, *Journal of Health Economics*, 27, 531-543.
- Tether, B.S. and Tajar, A. (2008), The organizational-cooperation mode of innovation and its prominence amongst European service firms, *Research Policy*, 37, 720-739.
- Tidd, J., Bessant, J. and Pavitt, K. (2001), Managing Innovation: Integrating Technological, Market and Organizational Change, 2nd edition, *Wiley*, Chichester, UK.
- Trencher, G., Yarime M., McCormick K. B., Doll C. N. H. and Kraines S.B. (2014), Beyond the third mission: Exploring the emerging university function of co-creation for sustainability, *Science and Public Policy*, 41, 151–179.

- Valero A. and Van Reneen, J. (2019), The economic impact of universities: Evidence from across the globe, Economics of Education Review, 68, 53-67.
- Vega-Jurado, J., Gutiérrez-Gracia, A. and Fernández-de-Lucio, I. (2009), Does external knowledge sourcing matter for innovation? Evidence from the Spanish manufacturing industry, *Industrial and Corporate Change*, 18, 637-670.
- Veugelers, R. (1997). Internal R&D expenditures and external technology sourcing, Research policy, 26, 303-315.
- Volberda, H. W., Van Den Bosch, F. A. and Heij, C. V. (2013). Management innovation: Management as fertile ground for innovation, European Management Review, 10, 1-15.
- Walker, R., Chen, J. and Aravind, D. (2015), Management innovation and firm performance: An integration of research findings, *European Management Journal*, 33, 407–422.
- Wooldridge J.M. (2014), Quasi-maximum likelihood estimation and testing for nonlinear models with endogenous explanatory variables, *Journal of Econometrics*, 182, 226-234.
- Wooldridge J.M. (2015), Control function methods in applied econometrics, *Journal of Human Resources*, 50, 420-445.

RECENTLY PUBLISHED "TEMI" (*)

- N.1380 An analysis of objective inflation expectations and inflation risk premia, by Sara Cecchetti, Adriana Grasso and Marcello Pericoli (July 2022).
- N.1381 The role of central bank communication in inflation-targeting Eastern European emerging economies, by Valerio Astuti, Alessio Ciarlone and Alberto Coco (October 2022).
- N. 1382 The heterogeneous effects of bank mergers and acquisitions on credit to firms: evidence from Italian macro-regions, by Silvia Del Prete, Cristina Demma, Iconio Garri, Marco Piazza and Giovanni Soggia (October 2022).
- N. 1383 Bank lending to small firms: metamorphosis of a financing model, by Paolo Finaldi Russo, Valentina Nigro and Sabrina Pastorelli (October 2022).
- N.1384 Macroeconomic effects of growth-enhancing measures in the euro area, by Alessandro Cantelmo, Alessandro Notarpietro and Massimiliano Pisani (October 2022).
- N.1385 *Revisiting the real exchange rate misalignment-economic growth nexus via the across-sector misallocation channel*, by Claire Giordano (October 2022).
- N. 1386 Labor market spillovers of a large plant opening. Evidence from the oil industry, by Matteo Alpino, Irene Di Marzio, Maurizio Lozzi and Vincenzo Mariani (October 2022).
- N. 1387 Unburdening regulation: the impact of regulatory simplification on photovoltaic adoption in Italy, by Federica Daniele, Alessandra Pasquini, Stefano Clò and Enza Maltese (October 2022).
- N.1388 *Liberalizing the opening of new pharmacies and hospitalizations*, by Andrea Cintolesi and Andrea Riganti (November 2022).
- N.1389 *Mediation Analysis Synthetic Control*, by Giovanni Mellace and Alessandra Pasquini (November 2022).
- N.1390 The effects of partial employment protection reforms: evidence from Italy, by Diego Daruich, Sabrina Di Addario and Raffaele Saggio (November 2022).
- N. 1391 *The role of majority status in close election studies*, by Matteo Alpino and Marta Crispino (November 2022).
- N.1392 Issuing bonds during the Covid-19 pandemic: is there an ESG premium?, by Fabrizio Ferriani (November 2022).
- N. 1393 *Economic fundamentals and stock market valuation: a CAPE-based approach*, by Maria Ludovica Drudi and Federico Calogero Nucera (November 2022).
- N. 1394 *The impact of "Metro C" in Rome on the housing market*, by Federica Daniele and Elena Romito (December 2022).
- N. 1395 *Gender quotas, board diversity and spillover effects. Evidence from Italian banks,* by Silvia Del Prete, Giulio Papini and Marco Tonello (December 2022).
- N.1396 *Real-time inequalities and policies during the pandemic in the US*, by Luisa Corrado, Daniela Fantozzi and Simona Giglioli (December 2022).
- N. 1397 The impact of sovereign tensions on bank lending: identifying the channels at work, by Fabiana Sabatini (December 2022).
- N. 1398 Make-up strategies and exchange rate pass-through in a low-interest-rate environment, by Alessandro Cantelmo, Pietro Cova, Alessandro Notarpietro and Massimiliano Pisani (December 2022).

^(*) Requests for copies should be sent to:

Banca d'Italia – Servizio Studi di struttura economica e finanziaria – Divisione Biblioteca e Archivio storico – Via Nazionale, 91 – 00184 Rome – (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

2020

- ALESSANDRI P. and M. BOTTERO, *Bank lending in uncertain times,* R European Economic Review, V. 128, WP 1109 (April 2017).
- ANTUNES A. and V. ERCOLANI, Public debt expansions and the dynamics of the household borrowing constraint, Review of Economic Dynamics, v. 37, pp. 1-32, WP 1268 (March 2020).
- ARDUINI T., E. PATACCHINI and E. RAINONE, *Treatment effects with heterogeneous externalities*, Journal of Business & Economic Statistics, v. 38, 4, pp. 826-838, **WP 974 (October 2014)**.
- BALTRUNAITE A., C. GIORGIANTONIO, S. MOCETTI and T. ORLANDO, Discretion and supplier selection in public procurement, Journal of Law, Economics, and Organization, v. 37, 1, pp. 134-166, WP 1178 (June 2018)
- BOLOGNA P., A. MIGLIETTA and A. SEGURA, *Contagion in the CoCos market? A case study of two stress events*, International Journal of Central Banking, v. 16, 6, pp. 137-184, **WP 1201 (November 2018).**
- BOTTERO M., F. MEZZANOTTI and S. LENZU, *Sovereign debt exposure and the Bank Lending Channel: impact on credit supply and the real economy*, Journal of International Economics, v. 126, article 103328, **WP 1032** (October 2015).
- BRIPI F., D. LOSCHIAVO and D. REVELLI, Services trade and credit frictions: evidence with matched bank *firm data*, The World Economy, v. 43, 5, pp. 1216-1252, **WP 1110 (April 2017).**
- BRONZINI R., G. CARAMELLINO and S. MAGRI, Venture capitalists at work: a Diff-in-Diff approach at latestages of the screening process, Journal of Business Venturing, v. 35, 3, WP 1131 (September 2017).
- BRONZINI R., S. MOCETTI and M. MONGARDINI, *The economic effects of big events: evidence from the Great Jubilee 2000 in Rome,* Journal of Regional Science, v. 60, 4, pp. 801-822, WP 1208 (February 2019).
- COIBION O., Y. GORODNICHENKO and T. ROPELE, *Inflation expectations and firms' decisions: new causal evidence*, Quarterly Journal of Economics, v. 135, 1, pp. 165-219, WP 1219 (April 2019).
- CORSELLO F. and V. NISPI LANDI, *Labor market and financial shocks: a time-varying analysis*, Journal of Money, Credit and Banking, v. 52, 4, pp. 777-801, **WP 1179 (June 2018).**
- COVA P. and F. NATOLI, *The risk-taking channel of international financial flows*, Journal of International Money and Finance, v. 102, **WP 1152 (December 2017).**
- D'ALESSIO G., *Measurement errors in survey data and the estimation of poverty and inequality indices,* Statistica Applicata - Italian Journal of Applied Statistics, v. 32, 3, **WP 1116 (June 2017).**
- DE BLASIO G., I. DE ANGELIS and L. RIZZICA, *Lost in corruption. Evidence from EU funding to Southern Italy*, Italian Economic Journal / Rivista italiana degli economisti, v. 6, 3, pp. 355–377, WP 1180 (December 2016).
- DEL PRETE S. and S. FEDERICO, *Do links between banks matter for bilateral trade? Evidence from financial crises*, Review of World Economic, v. 156, 4, pp. 859 885, WP 1217 (April 2019).
- D'IGNAZIO A. and C. MENON, *The causal effect of credit Guarantees for SMEs: evidence from Italy,* The Scandinavian Journal of Economics, v. 122, 1, pp. 191-218, **WP 900 (February 2013).**
- ERCOLANI V. and F. NATOLI, *Forecasting US recessions: the role of economic uncertainty*, Economics Letters, v. 193, WP 1299 (October 2020).
- MAKINEN T., L. SARNO and G. ZINNA, *Risky bank guarantees*, Journal of Financial Economics, v. 136, 2, pp. 490-522, WP 1232 (July 2019).
- MODENA F., E. RETTORE and G. M. TANZI, *The effect of grants on university dropout rates: evidence from the Italian case*, Journal of Human Capital, v. 14, 3, pp. 343-370, WP 1193 (September 2018).
- NISPI LANDI V., *Capital controls spillovers*, Journal of International Money and Finance, v. 109, WP 1184 (July 2018).
- PERICOLI M., *On risk factors of the stock–bond correlation,* International Finance, v. 23, 3, pp. 392-416, WP 1198 (November 2018).
- PIETRUNTI M. and F. M. SIGNORETTI, Unconventional monetary policy and household debt: the role of cashflow effects, Journal of Macroeconomics, v. 64, Article 103201, WP 1142 (October 2017).
- RAINONE E., *The network nature of OTC interest rates*, Journal of Financial Markets, v.47, article 100525, WP 1022 (July 2015).
- RAINONE E. and F. VACIRCA, *Estimating the money market microstructure with negative and zero interest rates*, Quantitative Finance, v. 20, 2, pp. 207-234, WP 1059 (March 2016).
- RIZZICA L., *Raising aspirations and higher education. Evidence from the UK's widening participation policy,* Journal of Labor Economics, v. 38, 1, pp. 183-214, **WP 1188 (September 2018).**
- RIZZICA L. and M. TONELLO, *Persuadable perceptions: the effect of media content on beliefs about corruption*, Economic Policy, v. 35, 104, pp. 679–737, WP 1043 (November 2016).

SANTIONI, R., F. SCHIANTARELLI and P. STRAHAN, *Internal capital markets in times of crisis: the benefit of group affiliation*, Review of Finance, v. 24, 4, pp. 773-811, WP 1146 (October 2017).

SCHIANTARELLI F., M. STACCHINI and P. STRAHAN, *Bank Quality, judicial efficiency and loan repayment delays in Italy,* Journal of Finance, v. 75, 4, pp. 2139-2178, **WP 1072 (July 2016).**

2021

- ACCETTURO A., A. LAMORGESE, S. MOCETTI and D. PELLEGRINO, *Housing Price elasticity and growth: evidence from Italian cities,* Journal of Economic Geography, v. 21, 3, pp. 367-396, WP 1267 (March 2020).
- AFFINITO M. and M. PIAZZA, Always look on the bright side? Central counterparties and interbank markets during the financial crisis, International Journal of Central Banking, v. 17, 1, pp. 231-283, WP 1181 (July 2018).
- ALBANESE G., E. CIANI and G. DE BLASIO, *Anything new in town? The local effects of urban regeneration policies in Italy*, Regional Science and Urban Economics, v. 86, **WP 1214 (April 2019)**.
- ALBANESE G., G. DE BLASIO and A. LOCATELLI, Does EU regional policy promote local TFP growth? Evidence from the Italian Mezzogiorno, Papers in Regional Science, v. 100, 2, pp. 327-348, WP 1253 (December 2019).
- ALBERTAZZI A., A. NOBILI and F. M. SIGNORETTI, *The bank lending channel of conventional and unconventional monetary policy*, Journal of Money, Credit and Banking, v. 53, 2-3, pp. 261-299, **WP 1094 (Jan 2017)**.
- ANZUINI A. and L. ROSSI, Fiscal policy in the US: a new measure of uncertainty and its effects on the American economy, Empirical Economics, v. 61, 6, pp. 2613-2634, WP 1197 (November 2018).
- APRIGLIANO V. and D. LIBERATI, Using credit variables to date business cycle and to estimate the probabilities of recession in real time, The Manchester School, v. 89, 51, pp. 76-96, WP 1229 (July 2019).
- AUER S., M. BERNARDINI and M. CECIONI, *Corporate leverage and monetary policy effectiveness in the euro area,* European Economic Review, v. 140, Article 103943, **WP 1258 (December 2019).**
- BANERJEE R, L. GAMBACORTA and E. SETTE, *The real effects of relationship lending*, Journal of Financial Intermediation, v. 48, Article 100923, **WP 1133 (September 2017).**
- BARONE G., F. DAVID, G. DE BLASIO and S. MOCETTI, *How do house prices respond to mortgage supply?*, Journal of Economic Geography, v. 21, 1, pp.127-140, **WP 1282 (June 2020).**
- BARONE G. and S. MOCETTI, *Intergenerational mobility in the very long run: Florence 1427-2011*, Review of Economic Studies, v. 88, 4, pp. 1863–1891, **WP 1060 (April 2016).**
- BARTOCCI A., L. BURLON, A. NOTARPIETRO and M. PISANI, Macroeconomic effects of non-standard monetary policy measures in the Euro Area: the role of corporate bond purchases, The Manchester School, v. 89, S1, pp. 97-130, WP 1241 (October 2019).
- BATINI N., A. CANTELMO, G. MELINA and S. VILLA, *How loose, how tight? A measure of monetary and fiscal stance for the euro area,* Oxford Economic Papers, v. 73, 4, pp. 1536-1556, **WP 1295 (September 2020).**
- BENETTON M. and D. FANTINO, *Targeted monetary policy and bank lending behavior*, Journal of Financial Economics, v. 142, 1, pp. 404-429, **WP 1187 (September 2018).**
- BUSETTI F., M. CAIVANO and D. DELLE MONACHE, Domestic and global determinants of inflation: evidence from expectile regression, Oxford Bulletin of Economics and Statistics, v. 83, 4, pp. 982-1001, WP 1225 (June 2019).
- BUSETTI F., M. CAIVANO, D. DELLE MONACHE and C. PACELLA, *The time-varying risk of Italian GDP*, Economic Modelling, v. 101, Article 105522, **WP 1288 (July 2020).**
- BUSETTI F., S. NERI, A. NOTARPIETRO and M. PISANI, Monetary Policy strategies in the new normal: a modelbased analysis for the Euro Area, Journal of Macroeconomics, v. 70, Article 103366, WP 1308 (December 2020).
- CAPOLONGO A. and C. PACELLA, Forecasting inflation in the Euro Area: countries matter, Empirical Economics, v. 61, 4, pp. 2477-2499, WP 1224 (June 2019).
- CARMIGNANI A., G. DE BLASIO, C. DEMMA and A. D'IGNAZIO, *Urbanization and firm access to credit,* Journal of Regional Science, v. 61, 3, pp. 597-622, **WP 1222 (June 2019).**
- CORNELI F., *Financial integration without financial development*, Atlantic Economic Journal, v. 49, 2, pp. 201-220, **WP 1120 (June 2017).**
- COVA P., P. PAGANO, A. NOTARPIETRO and M. PISANI, Secular stagnation, R&D, public investment and monetary policy: a global-model perspective, Macroeconomic Dynamics, v. 25, 5, pp. 1267-1287, WP 1156 (December 2017).
- DE PHILIPPIS M., *Multitask agents and incentives: the case of teaching and research for university professors,* Economic Journal, v. 131, 636, pp. 1643-1681, **WP 1042 (December 2015).**

- DEL PRETE S. and M. L. STEFANI, *Women as "Gold Dust": gender diversity in top boards and the performance of Italian banks*, Economic Notes, Monte dei Paschi di Siena, v. 50, 2, e12183, **WP 1014 (June 2015).**
- FERRERO G., M. LOBERTO and M. MICCOLI, *The assets' pledgeability channel of unconventional monetary policy*, Economic Inquiry, v. 59, 4, pp. 1547-1568, **WP 1119 (June 2017).**
- FIDORA M., C. GIORDANO and M. SCHMITZ, *Real exchange rate misalignments in the Euro Area,* Open Economies Review, v. 32, 1, pp. 71-107, **WP 1162 (January 2018).**
- GAMBACORTA L., G. RICOTTI, S. SUNDARESAN and Z. WANG, *Tax effects on bank liability structure*, European Economic Review, v. 138, Article 103820, WP 1101 (February 2017).
- HERTWECK M., V. LEWIS and S. VILLA, *Going the extra mile: effort by workers and job-seekers,* Journal of Money, Credit and Banking, v. 54, 8, pp. 2099-2127, **WP 1277 (June 2020).**
- LI F., A. MERCATANTI, T. MAKINEN and A. SILVESTRINI, *A regression discontinuity design for ordinal running variables: evaluating central bank purchases of corporate bonds,* The Annals of Applied Statistics, v. 15, 1, pp. 304-322, **WP 1213 (March 2019).**
- LOSCHIAVO D., *Big-city life (dis)satisfaction? The effect of urban living on subjective well-being*, Journal of Economic Behavior & Organization, vol. 192, pp. 740-764, **WP 1221 (June 2019).**
- LOSCHIAVO D., Household debt and income inequality: evidence from Italian survey data, Review of Income and Wealth. v. 67, 1, pp. 61-103, WP 1095 (January 2017).
- METELLI L. and F. NATOLI, *The international transmission of US tax shocks: a proxy-SVAR approach*, IMF Economic Review, v. 69, 2, pp. 325-356, **WP 1223 (June 2019).**
- NISPI LANDI V. and A. SCHIAVONE, *The effectiveness of capital controls*, Open Economies Review, v. 32, 1, pp. 183-211, WP 1200 (November 2018).
- PAPETTI A., Demographics and the natural real interest rate: historical and projected paths for the Euro *Area*, Journal of Economic Dynamics and Control, v. 132, Article 04209, **WP 1306 (November 2020).**
- PEREDA FERNANDEZ S., Copula-based random effects models for clustered data, Journal of Business & Economic Statistics, v. 39, 2, pp. 575-588, WP 1092 (January 2017).

2022

- ANDINI M., M. BOLDRINI, E. CIANI, G. DE BLASIO, A. D'IGNAZIO and A. PALADINI, Machine learning in the service of policy targeting: the case of public credit guarantees, Journal of Economic Behavior & Organization, v. 198, pp. 434-475, WP 1206 (February 2019).
- ANGELICO C., J. MARCUCCI, M. MICCOLI and F. QUARTA, Can we measure inflation expectations using twitter?, Journal of Econometrics, v. 228, 2, pp. 259-277, WP 1318 (February 2021).
- BARTOCCI A., A. NOTARPIETRO and M. PISANI, *Covid-19 shock and fiscal-monetary policy mix in a monetary union*, Economic challenges for Europe after the pandemic, Springer Proceedings in Business and Economics, Berlin-Heidelberg, Springer, WP 1313 (December 2020).
- BOTTERO M., C. MINOIU, J. PEYDRÒ, A. POLO, A. PRESBITERO and E. SETTE, *Expansionary yet different: credit supply and real effects of negative interest rate policy,* Journal of Financial Economics, v. 146, 2, pp. 754-778, **WP 1269 (March 2020).**
- BRONZINI R., A. D'IGNAZIO and D. REVELLI, *Financial structure and bank relationships of Italian multinational firms*, Journal of Multinational Financial Management, v. 66, Article 100762, **WP 1326 (March 2021)**.
- CANTELMO A., *Rare disasters, the natural interest rate and monetary policy,* Oxford Bulletin of Economics and Statistics, v. 84, 3, pp. 473-496, **WP 1309 (December 2020).**
- CARRIERO A., F. CORSELLO and M. MARCELLINO, *The global component of inflation volatility*, Journal of Applied Econometrics, v. 37, 4, pp. 700-721, **WP 1170 (May 2018).**
- CIAPANNA E. and G. ROVIGATTI, *The grocery trolley race in times of Covid-19. Evidence from Italy*, Italian Economic Journal / Rivista italiana degli economisti, v. 8, 2, pp. 471-498, **WP 1341 (June 2021).**
- CONTI A. M., A. NOBILI and F. M. SIGNORETTI, *Bank capital requirement shocks: a narrative perspective,* European Economic Review, v.151, Article 104254, **WP 1199 (November 2018).**
- FAIELLA I. and A. MISTRETTA, *The net zero challenge for firms' competitiveness*, Envnmental & Resource Economics, v. 83, pp. 85-113, **WP 1259 (February 2020).**
- GUISO L., A. POZZI, A. TSOY, L. GAMBACORTA and P. E. MISTRULLI, *The cost of steering in financial markets: evidence from the mortgage market,* Journal of Financial Economics, v.143, 3, pp. 1209-1226, **WP 1252 (December 2019).**
- LAMORGESE A. and D. PELLEGRINO, *Loss aversion in housing appraisal: evidence from Italian homeowners,* Journal of Housing Economics, v. 56, Article 101826, WP 1248 (November 2019).

- LI F., T. MÄKINEN, A. MERCATANTI and A. SILVESTRINI, *Causal analysis of central bank holdings of corporate bonds under interference*, Economic Modelling, v.113, Article 105873, WP 1300 (November 2020).
- LOBERTO M, A. LUCIANI and M. PANGALLO, *What do online listings tell us about the housing market?*, International Journal of Central Banking, v. 18, 4, pp. 325-377, **WP 1171 (April 2018).**
- MIRENDA L., M. SAURO and L. RIZZICA, *The economic effects of mafia: firm level evidence*, American Economic Review, vol. 112, 8, pp. 2748-2773, WP 1235 (October 2019).
- MOCETTI S., G. ROMA and E. RUBOLINO, *Knocking on parents' doors: regulation and intergenerational mobility,* Journal of Human Resources, v. 57, 2, pp. 525-554, WP 1182 (July 2018).
- PERICOLI M. and M. TABOGA, *Nearly exact Bayesian estimation of non-linear no-arbitrage term-structure models*, Journal of Financial Econometrics, v. 20, 5, pp. 807-838, **WP 1189 (September 2018).**
- ROSSI P. and D. SCALISE, *Financial development and growth in European regions*, Journal of Regional Science, v. 62, 2, pp. 389-411, WP 1246 (November 2019).
- SCHIVARDI F., E. SETTE and G. TABELLINI, *Credit misallocation during the European financial crisis,* Economic Journal, v. 132, 641, pp. 391-423, **WP 1139 (September 2017).**
- TABOGA M., Cross-country differences in the size of venture capital financing rounds: a machine learning approach, Empirical Economics, v. 62, 3, pp. 991-1012, WP 1243 (November 2019).

FORTHCOMING

- APRIGLIANO V., S. EMILIOZZI, G. GUAITOLI, A. LUCIANI, J. MARCUCCI and L. MONTEFORTE, *The power of textbased indicators in forecasting Italian economic activity*, International Journal of Forecasting, WP 1321 (March 2021).
- BARTOCCI A., A. NOTARPIETRO and M. PISANI, *Non-standard monetary policy measures in non-normal times,* International Finance, **WP 1251 (November 2019).**
- CIAPANNA E, S. MOCETTI and A. NOTARPIETRO, *The macroeconomic effects of structural reforms: an empirical and model-based approach*, Economic Policy, **WP 1303 (November 2022).**
- DI ADDARIO S., P. KLINE, R. SAGGIO and M. SØLVSTEN, *It ain't where you're from, it's where you're at: hiring origins, firm heterogeneity, and Wages, Journal of Econometrics,* WP 1374 (June 2022).
- FERRARI A. and V. NISPI LANDI, *Whatever it takes to save the planet? Central banks and unconventional green policy*, Macroeconomic Dynamics, **WP 1320 (February 2021).**
- FERRARI A. and V. NISPI LANDI, *Toward a green economy: the role of central bank's asset purchases,* International Journal of Central Banking, **WP 1358 (February 2022).**
- LILLA F., Volatility bursts: a discrete-time option model with multiple volatility components, Journal of Financial Econometrics, WP 1336 (June 2021).
- LOBERTO M, Foreclosures and house prices, Italian Economic Journal / Rivista italiana degli economisti, WP 1325 (March 2021).
- LOMBARDI M. J., M. RIGGI and E. VIVIANO, *Bargaining power and the Phillips curve: a micro-macro analysis, and wages,* Journal of the European Economic Association, WP 1302 (November 2020).
- RIZZICA L., G. ROMA and G. ROVIGATTI, *The effects of shop opening hours deregulation: evidence from Italy*, The Journal of Law and Economics, **WP 1281 (June 2020).**
- TANZI G. M., Scars of youth non-employment and labour market conditions, Italian Economic Journal / Rivista italiana degli economisti, WP 1312 (December 2020).